

Biosphärenregion Berchtesgadener Land

> Biosphärenreservat Rhön



Scientific results of the pilot project "Green Care | Nature & Mental Health" are now available

Strengthening mental health through mindfulness and relaxation activities in nature



<u>Translation:</u> dialecta - Zentr. f. Int. Sprachdienstleistungen GmbH www.dialecta.com Grünberger Str. 26 | 10245 Berlin 0800 550 551 552

Introduction

The Berchtesgadener Land Biosphere Reserve and the Rhön Biosphere Reserve were designated by UNESCO as model regions for sustainable development in 1990 and 1991 respectively and are part of a global network of over 700 biosphere reserves. Both biosphere reserves are characterised by particularly distinctive natural and cultural landscapes that symbolise the harmonious coexistence of man and nature. Their recreational value is essential for nature-based health tourism, as well as for the health-related use of the natural environment by local residents. In addition, both regions have an important tradition of spa treatment with famous spas and health resorts. Recreation and health are important local factors, as evidenced by the fact that both biosphere areas have been recognised as "Health Regions Plus" (Gesundheitsregionplus). One of the tasks of biosphere areas is to explore new ways of tackling societal and sustainability challenges and developing solutions. One such societal challenge is the increasing incidence of stress-related illnesses and depression, which not only places a financial burden on health care systems and businesses due to increased absenteeism, but also causes significant suffering and reduced quality of life for those affected. In order to help people suffering from depression and to support those looking for preventive measures against burnout and stress-related illnesses, the two Bavarian biosphere regions have combined their interdisciplinary expertise in the pilot project "Green Care – Nature and Mental Health" to scientifically study how guided nature experiences can benefit these target groups. Since November 2018, specific mindfulness and relaxation activities in nature have been developed, tested with groups in four-hour weekly sessions over a period of three to four weeks, and scientifically evaluated in order to provide recommendations on how these activities can be used meaningfully in the health sector in the future. Another question was whether these activities increased conservation behaviour. The focus was on patients with depression in the psychosomatic rehabilitation area and on individuals from the general population seeking to reduce stress and prevent burnout in the prevention area. Due to the high societal relevance of the research questions and the interdisciplinary approach, the Bavarian State Ministry for Health and Care [Bayerisches Staatsministerium für Gesundheit und Pflege] and the Bavarian State Ministry for the Environment and Consumer Protection [Bayerisches Staatsministerium für Umwelt und Verbraucherschutz] decided to jointly fund this project for the first time.

To ensure quality, the progress of the project and practical experiences were regularly discussed in an interregional project advisory working group. The working group included representatives of the governments of Upper Bavaria and Lower Franconia, the two ministries, various Bavarian universities, experts in spa medicine, the Bavarian State Chamber of Psychological Psychotherapists [Bayerische Landeskammer der Psychologischen Psychotherapeuten] and the Bavarian Association of Statutory Health Insurance Physicians [Kassenärztliche Vereinigung Bayern]. There was also a regular exchange at regional level with experts from the health and nature conservation area.

Independent scientific monitoring was provided from December 2019 to April 2023 by the evaluation team led by Prof. Dr. Elisabeth Kals, Chair of Social and Organisational Psychology at the Catholic University of Eichstätt-Ingolstadt. In addition to evaluating the group offerings by means of questionnaires, expert interviews rounded out the research design. The evaluation team presented its final report in May 2023. In addition to contributing to international research in the field of nature and health, the report is expected to provide insights for the practical implementation of nature-oriented health programmes in the local health sector of both biosphere regions and beyond.

Results

The results of the accompanying scientific research show that both patients with depression from rehabilitation clinics and healthy individuals from the general population seeking preventive measures for stress reduction benefited from activities in nature based on mindfulness and relaxation offerings. Analysis of the questionnaires showed that feelings perceived as beneficial were enhanced by the nature activities, while feelings perceived as stressful were alleviated through the offerings. This effect was still measurable three months after the group offerings, suggesting that the activities were integrated into the participants daily lives and had a positive and lasting effect. One reason for the ease of integration into daily life is that different types of natural and cultural landscapes were suitable for carrying out the activities. The same positive results were observed in remote forest areas as well as in smaller urban green spaces, wooded areas and meadows. In the clinical setting, patients with higher levels of depression on admission benefited most from the group offerings. The interviews with experts confirmed this finding. The results also indicate that the emotional connectedness to nature, which is a crucial prerequisite for the willingness to engage in nature conservation, increased as a result of the offerings. Accordingly, the study shows that the offerings provided are innovative and effective.

The research project is funded by the Bavarian State Ministry for Health and Care and the Bavarian State Ministry for the Environment and Consumer Protection with a total budget of EUR 868,000 and ran from November 2018 to September 2023.

Meike Krebs-Fehrmann

Green Care – Nature & Mental Health UNESCO Biosphere Reserve Berchtesgadener Land Meike.Krebs-Fehrmann@reg-ob.bayern.de



Department of Social and Organizational Psychology Evaluation team Joint project facilitation: Prof. Dr. Elisabeth Kals and PD Dr. Markus Müller Project colleagues: M. Sc. Patricia Zieris and M. Sc. Stefanie Bruckbauer

Green Care | Nature and Mental Health

Final report of the scientific monitoring study

The "Green Care | Nature and Mental Health" study is a scientific monitoring study commissioned by the administrative authorities of the UNESCO Biosphere Reserve Berchtesgadener Land and the UNESCO Biosphere Reserve Rhön, Bavarian part, as part of the "Green Care | Nature and Mental Health" project and awarded to the Chair of Social and Organisational Psychology at the Catholic University of Eichstätt-Ingolstadt. The scientific monitoring took place from December 2019 to April 2023.

Executive Summary

Background

Guided contact with nature has been shown to have positive effects on well-being. However, little is known about the feasibility and effectiveness of nature-based interventions in inpatient treatment for depression and their effects on mood, mental health, and environmental attitudes and engagement of healthy adults. A pilot was jointly launched by the UNESCO Biosphere Reserves Berchtesgadener Land and Rhön, Bavarian part to develop a model therapy and prevention offer that focuses on experiencing nature. The effectiveness of these mindfulness and relaxation-based nature interventions ("Greencare") as a supplement to psychosomatic rehabilitation treatment for depressive patients was investigated, and a group of healthy individuals were offered the mindfulness-based intervention. Treatments were compared to waitlist control groups. In the clinical setting, control groups received treatment in addition to the waitlist.

Methods

The study was divided in two settings: the clinical and the preventive setting.

In the clinical setting, inpatient psychosomatic rehabilitation inpatients with depression in two psychosomatic rehabilitation clinics were allocated either four sessions of a nature-based relaxation program (UNESCO Biosphere Reserve Rhön) or three sessions of a nature-based mindfulness training (UNESCO Biosphere Reserve Berchtesgadener Land) (Greencare; n = 116) or allocated to treatment as usual plus waitlist control group (TAU+WL, n = 111). TAU+WL patients received inpatient treatment plus a one- or two-day short intervention shortly before discharge. All patients received questionnaires at admission (T1) and before discharge (T2Greencare) or before the short intervention (T2TAU+WL). Greencare patients received follow-up questionnaires three months after the intervention (T3).

In the preventive setting, participants from the general population in the UNESCO Biosphere Reserve Berchtesgadener Land were allocated to three sessions of nature-based mindfulness training (Greencare, n = 84) or to a waitlist control group (WL, n = 49). Due to lockdown and health protection measures, 56 persons ultimately received the allocated Greencare intervention.

The main outcome was mood as assessed by the Positive and Negative Affect Schedule. Secondary outcomes were depression (clinical setting only), mindfulness, state self-compassion, nature-related mindfulness, emotional affinity towards nature, awareness of threats to nature, internal and external attribution of responsibility for the protection of nature, willingness to protect nature, nature conservation behavior, and contact with nature. Data were analyzed as intent-to-treat using mixed models repeated measures adjusting for propensity score and location.

Semi-structured expert interviews were conducted with 10 persons in the clinical setting and 7 persons in the preventive setting and analyzed using qualitative content analysis.

To evaluate the process of the interventions, questionnaires were completed after each session in the Greencare group, assessing well-being during the session and perceived effectiveness of intervention by the participants.

Results

In the clinical setting, group comparisons of the primary outcome between Greencare and TAU+WL revealed significant interactions of time*group, showing significantly greater increases of positive affect and greater decreases in negative mood in the Greencare group between T1 and T2. At follow-up (T3), the effects decreased by small effects in the Greencare group but remained significant compared to T1. Secondary outcomes analyses revealed significant interactions of time*group for self-compassion and emotional affinity towards nature and non-significant effects in the same direction for mindfulness. Groups did not differ in their amount of contact with nature. Subgroup analyses revealed more favorable follow-up effects on the primary outcomes for patients with higher levels of depression at T1 and with a childhood in a rural environment. Results showed that effects were independent of intervention location.

In the preventive setting, group comparisons of primary outcomes between Greencare and WL (T1 to T2) revealed significant interactions of time*group for the negative affect but not for positive affect. At follow-up, effects of negative affect decreased by small effects in the Greencare group but remained significant compared to T1. Secondary outcomes analyses revealed significant effects for nature-related mindfulness, emotional affinity towards nature, awareness of threats to nature, and internal and external attribution of responsibility for the protection of nature. These effects remained stable after three months. Subgroup analyses showed that positive affect improved particularly among individuals who had spent their childhood in urban settings.

Qualitative data from the interviews confirmed the effects of the interventions on psychological wellbeing, mental health, physical and social well-being. Experts rated the interventions a highly effective.

In the process evaluations, a large majority of the participants rated their well-being during the sessions and perceived effectiveness highly. The high ratings increased with the duration of the intervention.

Conclusions

Results provide evidence that participants in the clinical setting and healthy participants from the general population in the preventive setting benefit from a mindfulness- and relaxation-based nature intervention by improved affect, the main outcome variable. The effects are slightly reduced after three months. In the clinical setting, this reduction is smaller in patients with higher levels of depression at admission. Qualitative interviews with experts and process evaluation corroborate these effects. Results also show that in both settings, nature-related variables are positively influenced by the interventions. In the clinical setting, this is the case for emotional affinity towards nature, which is an important predictor of future conservation behavior. In the preventive setting, emotional affinity

towards nature and other nature conservation-relevant variables are positively influenced. This indicates that the interventions impact important predictors of nature-related behavior, even if they do not directly address conservation behavior. Overall, results show that the intervention was feasible and effective for patients and healthy individuals. The study Greencare - Nature and Mental Health was able to show that the realized offers are innovative and effective. A continuation of the project "GreenCare | Nature and Mental Health" is therefore expressly recommended from a scientific point of view.

Table of contents

Summa	iry II	
Executi	ve summary	J
Table o	f contents VI	1
List of f	igures IX	
List of t	ables XI	
1	SCIENTIFIC BACKGROUND	1
1.1	Influence of nature on biopsychosocial health	2
1.2	Nature-based interventions in therapy, rehabilitation and prevention	3
1.2.1	Nature-based interventions in therapy and rehabilitation	4
1.2.2	Nature-based interventions in prevention	6
1.3	Aims of the scientific monitoring study	9
1.4	Research questions	9
1.5	Principles of evaluation research10	D
2	METHODS10	D
2.1	Preconditions for the study1	D
2.2	Work schedule	1
2.3	Participants1	2
2.4	Data collection methods1	2
2.4.1	Pilot study 1	2
2.4.2	Questionnaire	3
2.4.3	Process evaluation 14	4
2.4.4	Interviews	5
2.5	Interventions1	7
2.6	Evaluation methods1	8
2.6.1	Questionnaire	8
2.6.2	Process evaluation	Э
2.6.3	Interviews	Э
2.7	Study quality20	D
3	CLINICAL TRIAL RESULTS	1
3.1	Description of the sample2	1
3.2	Main results22	2

3.2.1	Questionnaire
3.2.2	Process evaluation
3.2.3	Interviews
3.3	Clinical trial results by site
3.3.1	UNESCO Biosphere Reserve Rhön
3.3.2	UNESCO Biosphere Reserve Berchtesgadener Land
3.4	Subgroup analyses of questionnaire data
4	PREVENTION TRIAL RESULTS
4.1	Description of the sample
4.2	Main results
4.2.1	Questionnaire 40
4.2.2	Process evaluation
4.2.3	Interviews
4.3	Subgroup analyses
5	DISCUSSION OF THE RESULTS
5.1	Discussion of the results in relation to the research questions
5.2	Discussion of strengths and limitations54
6	Conclusion56
WORK	S CITED

List of figures

Figure 1	Planned timeline at the time of submission of application 1	11
Figure 2	Design of the Green Care study1	L7
Figure 3	Clinical trial flow chart	21
Figure 4	Positive and negative affect in the clinical trial: comparing the Green Care and TAU+WL groups from T1 (admission) to T2 (discharge)	23
Figure 5	Affect T1-T2-T3 in Green Care group of the clinical trial (T1-T2-T3)	23
Figure 6	Success-relevant project characteristics of a nature-based intervention in the clinical setting	25
Figure 7	Effects of the intervention on mental well-being in the clinical setting	26
Figure 8	Applicability of the intervention to diverse disorders in the clinical setting	27
Figure 9	Effects of the intervention on physical well-being in the clinical setting	28
Figure 10	Effects of intervention on social well-being in the clinical setting	29
Figure 11	Effects of the intervention on pro-environmental and nature conservation behaviour in the clinical setting	29
Figure 12	Assessment of the overall effectiveness of the intervention in the clinical setting	30
Figure 13	Determinants of individual-level effectiveness of the intervention in the clinical setting	30
Figure 14	Determinants of session-level effectiveness of the intervention in the clinical setting	31
Figure 15	Determinants of intervention effectiveness due to the natural environment in the clinical setting	32
Figure 16	Positive and Negative affect in the clinical study (UNESCO Biosphere Reserve Rhön): comparison between "Nature-based relaxation offering" group and TAU+WL group from T1 (admission) to T2 (discharge)	
Figure 17	Affect T1-T2-T3 in the clinical trial (UNESCO Biosphere Reserve Rhön)	34
Figure 18	Subjective well-being and subjective effectiveness of the intervention over time from the perspective of the clinical trial participants.	35
Figure 19	Positive and Negative affect in the clinical trial (UNESCO Biosphere Reserve Berchtesgadener Land): comparison between "Nature-based mindfulness training" group and TAU+WL group from T1 (admission) to T2 (discharge)	36
Figure 20	Affect T1-T2-T3 in the clinical study (UNESCO Biosphere Reserve Berchtesgadener	
	Land)	36

	Green Care – Final Report of the Scientific Monitoring Study
Figure 21	Flow chart of the prevention trial
Figure 22	Positive and Negative affect in the prevention trial: comparison of Green Care and WL group from T1 (before the first session) to T2 (after the last session)
Figure 23	Affect T1-T2-T3 in the prevention trial 41
Figure 24	Subjective well-being and subjective effectiveness of the intervention over the course of the intervention from the perspective of the participants in the prevention trial
Figure 25	Success-relevant project characteristics of a nature-based intervention in the preventive setting
Figure 26	Effects of the intervention on mental well-being in the preventive setting 44
Figure 27	Effects of the intervention on physical well-being in the preventive setting
Figure 28	Effects of the intervention on social well-being in the preventive setting
Figure 29	Effects of the intervention on pro-environmental and nature conservation behaviour in the preventive setting
Figure 30	Assessment of the overall effectiveness of the intervention in the preventive setting
Figure 31	Determinants of intervention effectiveness on the individual level in the preventive setting
Figure 32	Determinants of intervention effectiveness on the session level in the preventive setting
Figure 33	Determinants of the intervention effectiveness of the natural environment in the preventive setting

List of tables

Table 1	Characteristics of the clinical sample at baseline	22
Table 2	Age distribution of experts in the clinical trial	22
Table 3	Contact of experts with patients in the clinical trial	22
Table 4	Clinical trial experts' knowledge of intervention	22
Table 5	Overall evaluation of the offering by the experts in the clinical trial	22
Table 6	Characteristics of the prevention sample at baseline	39
Table 7	Age distribution of experts in the prevention trial	40
Table 8	Prevention trial experts' knowledge of intervention	40
Table 9	Overall evaluation of the offering by experts in the prevention trial	40

1 Scientific background

There can be no health without mental health. For individual citizens, mental health is a prerequisite for realising their intellectual and emotional potential and for finding and fulfilling their role in society.

(European Commission, 2005, p. 5)

The World Health Organisation (WHO) sees health not as the primary goal of life, but as a resource for everyday life (1986). Health is defined as "a state of complete physical, mental and social well-being and not merely the absence of disease" (WHO, 1948, p. 1). Therefore, by definition, the concepts of well-being and health are inherently interrelated. The biopsychosocial model base on Engel (1977) can be used to describe the complex relationships between these three facets of well-being. In this model, the functioning or health of an individual is understood as an interaction between physical, mental and social factors (ibid). Accordingly, health or well-being is seen as a dynamic biopsychosocial overall state of balance determined by multiple factors (Abraham et al., 2007; WHO, 1986).

With the onset of the COVID-19 pandemic, mental stress increased steadily (German Psychotherapeutic Association, 2021), and the percentage of people with moderate to severe symptoms of anxiety and depression rose sharply (Peters et al., 2020). However, the prevalence of mental illness had already risen significantly in the preceding decades. Mental illness is the most common cause of early retirement. Despite a dense network of treatment options for mental illnesses in Germany, patients have to wait an average of 19.9 weeks for a therapy spot (Bundespsychotherapeutenkammer, 2018). The associated sick days, sickness absence and reduced productivity are not only an individual burden for patients, but also have a huge socio-economic impact on society. Existing treatment options, such as psychopharmaceuticals or psychotherapy, consistently show moderate efficacy (Leucht et al., 2012; Munder et al., 2019), but many questions remain unanswered. Not all patients respond to treatment options, and in the case of depression, for example, long-term relapse is possible despite short-term success in therapy (Burcusa & Iacono, 2007).

There are many reasons for the increase in mental illness. On the one hand, an individual's genetic predisposition to cope with environmental factors and critical life events plays a role. On the other hand, today's achievement-oriented society with its high workload and stress levels has become a significant risk factor for mental illness (Ensinger, 2016; Marschall et al., 2020; Polz-Watzenig, 2020). Stress and a lack of work-life balance are the result. In addition, processes of urbanisation and digitalisation are leading to an increasing alienation of people from the natural environment (Heise & Hallermayr, 2022; Marschall et al., 2020). In addition, many forms of therapy only aim to treat symptoms (Thom et al., 2019). However, it is equally important to find a healthy approach to the demands of everyday life. Therefore, not only therapy and rehabilitation of mental disorders, but also prevention are becoming increasingly important.

1.1 Influence of nature on biopsychosocial health

Nature, i.e., the inanimate part of the natural environment (e.g., rocks, water and air) and the high biodiversity of flora and fauna (Hartig et al., 2014), is particularly relevant as an intervention setting in this context. The literature provides numerous explanatory approaches for how the mechanisms of experiences in nature affect human well-being. Three of the most prominent theories are presented below. Although they differ in their approaches, explanations and effect conditions, they all postulate a positive relationship between being in nature and well-being.

- Biophilia hypothesis (from ancient Greek: bios "life" and philia "love"; Kellert & Wilson, 1993; Wilson, 1984): According to this theory, humans have a genetically encoded emotional affinity for nature. It is therefore a human need to connect with nature physically, emotionally and cognitively. Accordingly, contact with nature is crucial for building and maintaining quality of life and well-being and can positively influence behaviour, cognition and psyche (Keniger et al., 2013).
- Attention restoration theory (Kaplan & Kaplan, 1989): Contact with nature helps to restore the attention capacity that has been depleted by the stress of everyday life and the associated mental fatigue and exhaustion. Spending time in nature thus induces a sense of relaxation.
- Psycho-evolutionary theory (Ulrich, 1983) or stress management theory (Ulrich et al., 1991): Being in nature and observing elements of nature leads to recovery from psychophysiological stress by influencing the limbic system and triggering an evolutionarily conditioned sense of safety. This effectively reduces stress over the long term (Bröderbauer, 2015).

The findings on the effects that contact with or being in nature have on human well-being have been studied extensively (Kals, 1998; Kals et al., 2023). According to Cox et al. (2017) and Cervinka et al. (2014), nature has a high healing potential and can affect well-being in a preventive and rehabilitative way (Abraham et al., 2007; Bowler et al., 2010; Flade, 2018). In line with the three pillars of the biopsychosocial model, selected empirical findings are briefly outlined below.

- Experiences in nature and mental well-being: Experiences in nature can promote mental well-being by, among other things, having nature serve as a place of relaxation. As a result, they can reduce mental exhaustion (Berman et al., 2008; Berto, 2005) and improve concentration (Bratman et al., 2015; Hartig et al., 2003; Tennessen & Cimprich, 1995) and frustration tolerance (Cackowski & Nasar, 2003; Kuo & Sullivan, 2001). In contact with nature, positive emotions can be generated and negative emotions reduced (Ballew & Omoto, 2018; Bowler et al., 2010; McMahan & Estes, 2015; Russell et al., 2013). Likewise, experiences in nature decrease the occurrence of mental illnesses (Beyer et al., 2014; Cervinka et al., 2014; Cox et al., 2017).
- Experiences in nature and physical well-being: Experiences in nature can promote physical well-being at many levels, including by encouraging movement, relaxation and a sense of oneness with nature. Empirical studies show that contact with nature can reduce pain (Ulrich,

1984), discomfort (Kaplan & Kaplan, 1989) and stress (Ulrich et al., 1991) and improve sleep (Morita et al., 2011). It can also boost the immune system (Kuo, 2015).

 Experiences in nature and social well-being: Experiences in nature can improve social wellbeing, especially when shared with other people. Being in nature provides opportunities for social encounters (Coley et al., 1997; Kals et al., 1999; Maas et al., 2009), during which social interaction (Astles, 2015; Kweon et al., 1998; Nicolè & Seeland, 1999) and social skills can be promoted (Annerstedt & Währborg, 2011; Weinstein et al., 2009). As a result, social relationships can be established and feelings of isolation reduced (Kingsley & Townsend, 2006; Maas et al., 2009).

In summary, based on these selected findings, it can be said that when opportunities for experiences in nature are created, well-being in all its dimensions and thus biopsychosocial health can be promoted (Kals & Nisbet, 2019; Zieris et al., 2023). Furthermore, environmentally relevant behaviour is strongly influenced by emotions (Kals & Müller, 2012). Thus, emotional attachment to nature can promote proenvironmental and nature conservation attitudes and behaviour, which can also be promoted by positive experiences in nature (Müller et al., 2009; Müller et al., 2013). In this sense, experiences in nature can both help protect health and serve sustainability goals (Bruckbauer et al., 2022; Müller et al., 2014).

1.2 Nature-based interventions in therapy, rehabilitation and prevention

Given the theoretical assumptions presented and the existing empirical evidence on the relationship between nature and health, it is logical to utilise the diverse effects of contact with nature in the prevention, therapy and rehabilitation of mental disorders. Nature-based interventions, i.e., guided contact with nature, represent an approach to integrate the positive effects of experiences in nature into a psychotherapeutic framework (Annerstedt & Währborg, 2011). These interventions incorporate not only individual experiences but also the natural environment into the therapeutic process (Dienemann, 2020; Petzold et al., 2019). This approach adopts a holistic perspective with a focus on salutogenesis and prevention, using a method based on experience, resources and actions rooted in the tradition of the humanistic worldview and client-centred psychotherapy according to Rogers (1981). Nature-based interventions include forest bathing (Japanese: shinrin-yoku), forest therapies, garden therapies, forest education and Green Care. Green Care is a broad term that encompasses all approaches that contribute to maintaining and enhancing human well-being by involving nature (animals, plants, gardens, forests and landscapes) (Haubenhofer et al., 2010; Steigen et al., 2016).

Empirical research on the effectiveness of nature-based interventions is still in its early stages. However, there is evidence that even a single session in nature can have diverse effects on biopsychosocial well-being (Shanahan et al., 2019). In the review by Coventry et al. (2021), which examined the effectiveness of nature-based interventions for people with or without mental or physical health problems, including 50 studies, these interventions led to significant improvements in depressive mood, positive and negative affect, and anxiety. The most effective interventions were delivered over a period of 8 to 12 weeks, with an optimal duration of 20 to 90 minutes. In particular,

the combination of mindfulness activities and experiences in nature appears to have a positive impact on well-being (Djernis et al., 2019). The following is a review of recent research on the effectiveness of nature-based interventions. First, studies in the context of therapy and rehabilitation are presented, followed by those in the context of prevention.

1.2.1 Nature-based interventions in therapy and rehabilitation

The study by Bielinis et al. (2019) investigated the impact of forest bathing on the mental health of **inpatients with affective or psychotic disorders**. Patients were invited to participate in one-hour forest walks in groups of four to five people under the supervision of qualified therapists. Data were collected using questionnaires before and after the intervention. Patients with affective disorders showed improvements in mood on various mood scales, and their anxiety levels also decreased significantly. Patients with psychotic disorders also experienced a reduction in anxiety and an increase in vitality.

The meta-analysis by Grassini (2022) analysed studies on the effects of walks in nature on **people with depression and anxiety**. Six studies published between 2013 and 2020 were identified and met the inclusion criteria of pre- and post-testing and experimental and control conditions. The data showed that walks in nature were effective in improving mental health.

Keenan et al. (2021) also focused on **people with depression and/or anxiety** (n = 50). In an experimental design, participants were randomised to either the experimental group (walk in nature with the task of noticing three good things in nature (TGTiN)) or the control group (walk in the city). Both groups were instructed to walk for 30 minutes every day for five consecutive days. In addition to the data collection at the beginning and end of the five days, further data were collected six weeks later. In the experimental group, connectedness to nature and positive affect increased significantly more than in the control group in both the post-intervention and follow-up assessments. Negative affect decreased in the experimental condition, while well-being during the walk in nature was significantly greater.

The study by Hyvönen et al. (2023) also investigated the effectiveness of a nature-based intervention in **patients with depression**. Participants were randomised to the experimental group (n = 59, one nature-based session per week for twelve weeks in addition to standard treatment) or the control group (n = 77, treatment as usual). Pre-post measurements show that reductions in mental stress and increases in restorative experiences were greater in the experimental group than in the control group. Depression scores decreased equally in both groups.

Joschko et al. (2023) investigated the relationship between nature-based therapy, mental health and connectedness to nature **in patients with psychosomatic disorders** (n = 19). Over a four-week period, patients attended one-hour nature-based sessions three times a week. Data were collected before the first and after the last session. Significant improvements in mental well-being and connectedness to nature were reported, and participants in the study subjectively perceived the intervention as effective.

The target group of the study by Corazon et al. (2018) consisted of **individuals diagnosed with binge eating disorder** (n = 20). The aim was to compare the effectiveness of a nature-based intervention with

self-help group meetings. The nature-based intervention took place once a week for three hours over twelve weeks, while the self-help groups met once a week for three hours over ten weeks. The mixedmethods approach shows that both actions led to an increase in overall mental well-being. However, participants found it easier to integrate the content of the nature-based intervention into their daily lives than the content of the self-help group meetings.

Choi et al. (2021) investigated the effects of mindfulness-based mandala colouring (MBMC) in nature on **people with chronic pain** (n = 36). Participants were randomly assigned to the experimental group (colouring) or the control group (city tour). Data was collected before and after the intervention. Participants in the experimental group showed significant improvements in depressive symptoms, anger and stress levels.

Han et al. (2016) also investigated the efficacy of nature-based interventions for **individuals with chronic pain** (n = 61). Participants were non-randomly assigned to an experimental group (two-day forest therapy programme) or a control group (no intervention). Data were collected before and after the study. Participants in the experimental group reported a significant reduction in pain and depression and a significant improvement in health-related quality of life. They also showed improvements in physiological parameters compared with the control group.

The study by Chun et al. (2017) focused on **patients with chronic stroke** (n = 59). It investigated the effects of forest therapy on depression and anxiety. Participants in the study were randomly assigned to the experimental group (four days of forest therapy) or the control group (the same programme as the experimental group in an urban setting). Data was collected before and after the intervention. The reduction in depression and anxiety was significantly greater in the experimental group than in the control group.

A review (Britton et al., 2020) systematically assessed the effectiveness of therapeutic interventions in the setting of outdoor water environments (Blue Care) on health and well-being. Thirty-three studies published between 2004 and 2017 were included, evaluating interventions for **individuals with** (psycho)therapeutic needs. The studies suggest that Blue Care can directly benefit mental health and psychosocial well-being. The relationships between Blue Care and various indicators of health and well-being are positive, but weak.

1.2.2 Nature-based interventions in prevention

Gittins et al. (2023) used a mixed methods approach to demonstrate that even a single nature-based intervention increases the likelihood of subsequent independent visits to a forest. They conducted a longitudinal questionnaire study (n = 120) and focus groups (n = 20). According to the authors, **nature-based interventions can serve as a "turning point"** (p. 4) for engaging with nature and reducing mental, emotional and socio-cultural barriers to spending time in nature.

In their review, Corazon et al. (2019) examined the **psychophysiological effects** of nature-based interventions on individuals' stress management. They included 36 studies that were published between 2010 and 2018. The results suggest that spending time outdoors and in nature has a positive

effect on various emotional parameters related to stress reduction. When it came to physiological effects, the results were mixed.

In their study, Martin et al. (2020) examined the relationships between different types of contact with nature, connectedness to nature and **subjective well-being**. They showed that only direct contact with nature (at least once a week) was associated with general health and environmentally friendly behaviour; however, spending time in neighbourhood green spaces or watching nature documentaries was not sufficient to produce this effect (n = 4,960). Emotional connectedness to nature moderated these connections. The authors concluded that interventions need to promote both contact with nature (nature-based interventions) and emotional connectedness to improve well-being.

Bailey and Kang (2022) examined the extent to which physical activity during a nature-based intervention influenced **cognitive performance and mental health** in a sample of 50 US students. Students were randomly assigned to a walking group (ten-minute walk in an adjacent park) or a rest group (ten minutes of sitting in an adjacent park). Cognitive performance was assessed using the Stroop test ten minutes before and after the intervention. Relaxation, attention and motivation were measured using EEG during the ten-minute intervention. Mindfulness and connectedness to nature were self-reported. The study found that cognitive performance improved equally in both groups. The level of relaxation during the intervention was the best predictor of later performance. Participants with high connectedness to nature and mindfulness showed less evidence of rumination. Finally, the walking group had higher levels of relaxation during the intervention than the sitting group.

In a proof-of-principle study, Owens et al. (2020) investigated the effectiveness of a nature-based intervention on **mental well-being**. Students were randomly assigned to one of three conditions: 20 minutes of meditation in nature (n = 23), 20 minutes of indoor meditation (n = 22), or an active control group (n = 23). Subjective depressive symptoms and well-being were assessed after the intervention and two weeks later. While depressive symptoms decreased in both meditation groups, well-being improved in the long term only in the nature-based intervention.

Huber et al. (2023) took a similar approach, investigating the effects of two nature-based interventions (forest therapy and mountain hiking) on **the mental and physical health** of couples with a sedentary lifestyle. Both interventions ($n_{\text{forest therapy}} = 23$; $n_{\text{mountain hiking}} = 22$) were carried out for three to four hours a day for one week. Data were collected at three different time points (before the intervention, on the seventh day of the intervention, and 60 days after the intervention). Both interventions led to improvements in the subjective mood and life satisfaction, as well as reductions in blood pressure.

In their randomised controlled study, Daniels et al. (2022) investigated the effects of a nature-based intervention on **mental health and perceived stress at work**. The experimental group (n = 25) participated in nature-based activities twice a week for two hours each time over the course of three consecutive weeks. The control group (n = 20) received no intervention. Analyses show that participants in the experimental group had lower levels of burnout and salivary cortisol and higher visual information processing speeds after the intervention as compared to those in the control group. Selective attention also improved in the experimental group. The authors concluded that nature-based interventions during work can reduce stress and improve cognitive performance.

The relationship between contact with nature and **mental well-being at work** was also investigated by Ho et al. (2022). A total of 90 university employees were randomly assigned to an experimental group (n = 48) or a control group (n = 42). Participants in the experimental group completed a 30-minute nature-based intervention during their lunch break on ten consecutive weekdays. The self-reported mental stress (anxiety, depression, and perceived stress) and mental well-being (including positive affect) were collected at the beginning of the study, after the tenth session, and three months later. Significant interaction effects of the intervention on mental stress and well-being were found, although these effects were no longer significant after three months.

In their randomised study, Oafsdottir et al. (2020) investigated the effects of nature-based interventions on the **mood and psychophysiological responses to stress**. Thirty Icelandic students participated in a roughly 40-minute intervention: walking in nature, watching nature on television, or running on a treadmill in a gym. Data were collected before and after the intervention, as well as after artificially induced stress. All the interventions led to a reduction in the stress hormone cortisol, with the greatest reduction observed after walking in nature. The mood of the participants in this group also improved more than in the other two groups.

The randomised controlled study by Ma et al. (2022) focused on the influence of mindful walking in nature on **sleep quality, mood and mindfulness** in British students. Participants were randomly assigned to the experimental group (n = 52; walking in a natural environment) or the control group (n = 52; walking in an urban environment). Both groups walked for 35 minutes each day for seven consecutive days and were surveyed on their subjective assessments at four time points (pre-intervention, after the first and last walk, and five days after the last walk). The participants in both groups reported improvements in mindfulness, sleep quality and mood.

In their study, Irvine et al. (2020), on the one hand, investigated the effects of a nature-based intervention on **mental and physical well-being** and, on the other, explored ways to increase motivation and commitment to participate in such interventions. Thirteen people participated in a twelve-week walking group. Data were collected by means of self-reporting and activity trackers for a pre-post comparison. After completing the intervention, participants reported improved sleep, more positive emotions, increased calmness and better concentration. An objective increase in physical activity was also observed.

While many studies of forest bathing (shinrin-yoku) focus on the physical effectiveness of the intervention, the studies by Kotera and colleagues investigated its effects on **mental well-being**. Kotera and Fido (2022) investigated the effectiveness of a three-day shinrin-yoku seminar on the mental well-being of Japanese students (n = 25). Participants' self-reports were collected immediately before, immediately after, and two weeks after the intervention, and included measures of mental well-being, self-compassion, and mindfulness. While mental well-being did not change, significant improvements were found in self-compassion and mindfulness. In addition, the review by Kotera et al. (2022) provided evidence that different elements of forest bathing (breathing, walking, yoga) can effectively reduce mental symptoms in the short term. This analysis included 20 studies published before November 2019 that aimed to reduce depression, anxiety and stress. Only longitudinal studies were

included, although most of them only included pre-post measurements and did not have follow-ups to examine long-term effects.

1.3 Aims of the scientific monitoring study

The evaluation research in the "Green Care | Nature and Mental Health" project pursues three aims:

- To verify the expected goals of the intervention, which can be derived from the theoretical foundations and previous empirical evidence.
- To fill the current gaps in scientific knowledge about the effectiveness of nature-based interventions, considering both therapeutic and rehabilitative settings as well as the preventive area. The focus is on human mood, with a secondary focus on the psychobiosocial health of individuals as a whole, as well as on pro-environmental and nature conservation attitudes and behaviour.
- To contribute to the strengthening of the reputation of nature-based interventions and the spread of nature conservation values.

1.4 Research questions

The scientific evaluation pursues two overarching research questions:

- How does nature affect individuals, and to what extent do nature-based interventions influence participants' mental health and attitudes towards nature? How do participants subjectively perceive and evaluate these effects? Can positive effects on health- and naturerelated variables be demonstrated through an experimental scientific design?
- Are there long-term effects on their mental health and increased nature-related consciousness and behaviour?

Different samples, measurement methods and time points are used to answer two central research questions:

- Is the intervention effective in the short term and is it subjectively perceived as effective by the participants?
- Is the intervention effective in the long term, and do participants integrate what they have learned into their daily lives?

1.5 Principles of evaluation research

The study by the external evaluation team is based on the commitment to scientific independence. The impact evaluation is also committed to the four fundamental attributes and the corresponding

regulations established by the German Society for Evaluation (DeGEval - Gesellschaft für Evaluation e.V., 2016) as standards for good evaluation:

- Utility: Alignment with clarified evaluation purposes and related information needs
- Feasibility: Realistic, well thought-out, diplomatic and cost-conscious planning and implementation
- Fairness: Respectful and fair treatment of all stakeholders
- Accuracy: Producing valid information and results

2 Methods

2.1 Preconditions for the study

The study was approved by the Ethics Committee of the Catholic University of Eichstätt-Ingolstadt (No. 029-2020 on 26 August 2020). It has also been registered in the German Clinical Trials Register (Trial number DRKS00023369). The Universal Trial Number (UTN) registered with the World Health Organisation (WHO) is U1111-1260-7305.

All participants were informed about the aims of the study and gave written consent to participate in the study and to use their data.

2.2 Work schedule

An overview of the originally planned work schedule is shown in Figure 1. This figure illustrates the timeline as it was drawn up by the evaluation team at the time of submission of the application. During the course of the project, it was adapted to the prevailing circumstances in consultation with the responsible parties from the two UNESCO biosphere reserves.



Over time (42 months)

Figure 1. Planned timeline at the time of submission of application

The evaluation team carried out the following tasks in detail:

- Conducting a literature review and clarifying rights of use
- Designing the survey instruments
- Conducting the pretest
- Developing other key documents
- Drafting the ethics application for the evaluation of the intervention and submitting it to the Ethics Committee of the Catholic University of Eichstätt-Ingolstadt (after the description of the intervention had been prepared by the staff responsible for interventions in UNESCO biosphere areas)
- Coordinating with the project sites and continuous communication throughout the project
- Registering the study (drks.de)
- Supervising data collection
- Conducting data collection at the T3 follow-up point
- Conducting interviews
- Data processing
- Data analysis
- Presenting and discussing interim and final results at meetings of the Project Advisory Working Group (PAG)
- Writing the final report
- Writing scientific publications

2.3 Participants

Inclusion criteria: Participants of either gender could be included in the study.

For the clinical setting: Inclusion criteria consisted of depressive disorders (diagnosed by an initial medical assessment on admission to the clinic), adequate tetanus vaccination status, the presence of sturdy footwear and weatherproof clothing, and knowledge of the German language. The minimum age was set at 18 and the maximum age was initially set at 59. Due to the circumstances in the clinics, older people could also participate in the study in consultation with the medical staff.

For the preventive setting (general population): Inclusion criteria consisted of an age between 18 and 59 years, adequate tetanus vaccination status, possession of sturdy footwear and weatherproof clothing, and knowledge of the German language. Again, the upper age limit was raised in consultation with the intervention coordinators due to special circumstances.

Exclusion criteria consisted of manic episodes, delusional and other severe psychiatric comorbidities, physical impairments that limited mobility in the field, and inability to give informed consent. For the clinical setting, the selection of eligible patients for the offering was determined by the admitting senior physicians and case managers (UNESCO Biosphere Reserve Rhön) or the chief physician, senior physicians and/or case therapists (UNESCO Biosphere Reserve Berchtesgadener Land).

2.4 Data collection methods

2.4.1 Pilot study

In order to assess the quality criteria of the questionnaire instruments used, they were tested in a pilot study. An ad-hoc sample of *N* = 205 participants from the general population was recruited to complete an online questionnaire between 26 February and 4 April 2020. The data were then analysed using factor analyses, reliability analyses, correlation analyses, multiple regression analyses and mean comparisons using t-tests. The program SPSS Statistics 29 (IBM Corp., Armonk, NY, USA) was used for statistical analyses. The results confirmed the suitability of most of the scales for the project. The insufficient results prompted the decision to measure the constructs of social support and compassion using well-validated scales, specifically the ENRICHD Social Support Inventory (ESSI) (Cordes et al., 2009) for social support and the State Self-Compassion Scale Short Form (SSCS-S) (Neff et al., 2021) for compassion. As both scales already had data supporting their validity, further validation was not pursued within the project. In addition, the project team decided to use a measure of depression in the clinical sample (Patient Health Questionnaire (PHQ-9); Kroenke et al., 2001).

2.4.2 Questionnaire

Primary endpoint

The primary endpoint was the assessment of positive and negative affect using the *Positive and Negative Affect Schedule Short Form* (PANAS-SF; Mackinnon et al., 1999; Watson et al., 1988). The participants were asked to rate their current mood for five positive and five negative adjectives on a five-point Likert scale ranging from 1 (very low or not at all) to 5 (extreme). The means of the items were aggregated into Negative affect and Positive affect subscales.

Secondary endpoints

Depression was measured exclusively in the clinical setting using the nine-item *Depression Module of the Patient Health Questionnaire (PHQ-9)* (Kroenke et al., 2001). Each item is rated on a four-point Likert scale ranging from 0 (not at all) to 3 (almost every day). The total score was used as a continuous variable in the analyses of results. In addition, a cut-off of 0 to 9 (no or mild depression) and 10 to 27 (probable major depressive disorder) was applied (Meanea et al.; 2012) to differentiate between patients with low and high distress.

The other secondary endpoints were used in both the clinical and preventive settings.

Mindfulness was measured using the *Mindfulness Attention Awareness Scale (MAAS)* (Brown & Ryan, 2003). It consists of 15 items scored on a scale from 1 (almost never) to 5 (almost always) (sample item: "I could have a feeling and not realise it until some time later"). A total mean score was calculated from all items for further analysis.

Self-compassion was measured as a momentary state using the *State Self-Compassion Scale Short Form* (*SSCS-S*; Neff et al., 2021). The short form consists of six items (sample item: "I give myself the care and understanding that I need"). Responses were rated on a scale from 1 (does not apply to me at all) to 5 (applies to me quite a lot). The total score was calculated as the mean of all items.

Social support was measured using the *ENRICHD Social Support Inventory* (*ESSI*; Cordes et al., 2009), which consists of six items (sample item: "When you need to talk, is there someone who really listens to you?"). These items were rated on a five-point scale from 1 (does not apply to me at all) to 5 (applies to me a great deal).

Nature-related mindfulness was collected with five items (sample item: "In nature, I pay attention to sensations such as the sun on my face or the wind in my hair"; Wastlhuber, 2019). These and all subsequent items were measured on a six-point scale ranging from 1 (does not apply at all) to 6 (applies completely).

Emotional connection to nature (Kals et al., 1999; Wastlhuber, 2019) was measured with six items (sample item: "In nature I become more relaxed and feel uplifted").

Awareness of environmental threats (Blum, 2019) was assessed with three items (sample item: "If nothing fundamental changes, environmental threats will get worse in the coming years").

Attribution of responsibility for nature conservation (Blum, 2019) was measured with four items, with two items each for internal (sample item: "I feel responsible for actively contributing to nature conservation myself") and external responsibility (sample item: "The government is responsible for doing something about nature conservation").

Nature conservation attitudes (Blum, 2019; Wastlhuber, 2019) were assessed with five items (sample item: "I am generally willing to draw attention to the importance of nature conservation").

Nature conservation behaviour (Blum, 2019; Wastlhuber, 2019) was measured with three items (sample item: "In my everyday life, I generally act consistent with nature conservation").

In addition, single items were included to measure *childhood contact with nature* ("I spent most of my childhood and adolescence in an urban area") and *current contact with nature* ("I currently spend a lot of time in nature").

The questionnaire also includes additional questions on *demographic information* (gender: male, female, mixed; age in years; children yes/no) and *previous experience with training in nature or mindfulness training* (yes/no, and a free-response option to specify the context in which this training took place). Finally, participants were asked to provide a self-generated *code* that would allow the questionnaires to be matched in the case of multiple measures.

2.4.3 Process evaluation

In order to complement the summative evaluation with questionnaires by including data directly resulting from the interventions, individual evaluations were used as process evaluation instruments. This took the form of a one-page questionnaire completed by the facilitators after each session. Providing the group number assigned by the evaluation team made it possible to clearly assign individual evaluations to sites and groups. The instrument is divided into two sections: a) situational aspects and b) evaluation of the session by the intervention participants.

Situational aspects of the session included the location (name of the site and description of the natural environment), time (date and time), group size and weather conditions. Any special aspects of the session could also be noted.

For the participants' *evaluation of the session*, they were asked by the facilitator to rate their wellbeing during the session on a four-point Likert scale (1 = I did not feel comfortable to 4 = I felt very comfortable). They were also asked to rate the perceived effectiveness of the intervention session (1 = I did not find it effective to 4 = I found it very effective). The nature of the questions was left to the discretion of the facilitator.

The process evaluation instrument was used in all experimental groups (clinical trials at both sites and prevention trial). In addition, a process evaluation of the clinical control groups was carried out in the UNESCO Biosphere Reserve Rhön.

2.4.4 Interviews

Another component of the empirical data collection consisted of semi-structured, guideline-based interviews conducted with experts. Experts are defined as individuals who possess knowledge of the process, interpretation, practice and action in a specific context (Bogner & Menz, 2002). Their knowledge can be made useful for real-life circumstances and thus serve as recommendations for action in other instances with the same action context, in this case nature-based intervention.

The selection of interviewees was made in consultation with the staff of the two UNESCO biosphere areas. The target number of interviews to be conducted was based on the number of people involved and their willingness to participate in the interviews. The consent of the experts to participate in the interview and to record the conversation for later detailed data analysis was obtained in writing in advance and confirmed verbally at the beginning of the interview.

Two guidelines were developed by the evaluation team for conducting semi-structured interviews: a) a guideline for the clinical study (the names "UNESCO Biosphere Reserve Rhön" or "UNESCO Biosphere Reserve Berchtesgadener Land" were used where appropriate), and b) a guideline for the prevention trial. All interviews began with a welcome message explaining the purpose of the interview and the formalities (including assurances of anonymity and information on data protection).

General information about the person was then collected, including their role or function in the clinic (clinical trial) or within the intervention (prevention trial) and the age group of the experts (under the age of 30, between the age of 30 and 40, between the age 40 and 50, between the age of 50 and 60, above the age of 60). In both studies, knowledge of the intervention was assessed on a four-point Likert scale (1 = I know absolutely nothing about it to 4 = I am familiar with the offering). In the clinical trial, an additional four-point Likert scale (1 = I have almost no contact with patients to 4 = I have a lot of contact with patients) was used to assess how much contact respondents had with patients.

The content part of the interviews, based on five psychological constructs, was then initiated, and for each construct primary questions were formulated and presented to all interviewees. In addition, the guidelines included construct-specific secondary questions that could be used if the primary question posed to the interviewee provided limited information or if there was uncertainty. The wording of the

questions on constructs in the guidelines for the clinical trial and the prevention trial differed only with regard to the use of "patients" and "participants".

For the construct of *interest in and popularity of the intervention*, respondents were asked about the extent to which the intervention was accepted by the target group (clinical trial: patients and clinic staff; prevention trial: participants) from their perspective.

In order to assess the *organisation and quality of the offering*, the interviewees were asked about the availability and transparency of information on the intervention, as well as their assessment of the cooperation between the intervention facilitator and the interviewee and the patients (clinical trial) or participants (prevention trial). There was also an opportunity to make suggestions for improvement regarding the quality of the offering and the cooperation.

The central construct was the *effectiveness of the offering on health-related variables*, which was discussed in the third part of the guideline. This part of the interview explored the effectiveness of the intervention on the biopsychosocial health of the intervention participants. Follow-up questions focused on short- and long-term effects and possible differences in the effectiveness of the intervention between participants.

The fourth construct was the *importance of the location of the intervention*. This asked how much the natural environment in which the intervention took place was important for its effectiveness. Secondary questions related to whether and to what extent participants made a connection between health and environmental protection, and whether participants expressed any related experiences during the nature-based interventions.

The construct relating to the *effectiveness of the offering on environmental variables* focused on the impact of the intervention on participants' pro-environmental and nature conservation mindset and attitudes. Questions going into more depth focused on whether participants discussed nature or the particular UNESCO biosphere area during or after the intervention, or whether the intervention led to greater engagement with environmental issues.

The interviews concluded with an *overall evaluation of the offering*, where interviewees were asked to rate the effectiveness of the intervention on a four-point Likert scale (1 = I do not find it very effective to 4 = I find it very effective).

Before concluding the interview, the experts were given the opportunity to provide any additional information that had not been covered.

In addition to these data, a postscript was prepared for each interview to provide information not recorded on the tape for later analysis. This included the date and duration of the interview, situational aspects of the interview, relevant topics of conversation before and after the audio recording, and the focus and characteristics of the interview.

All the measurement instruments used in the study can be found in Annex A.

2.5 Interventions

The study was divided into four arms, two of which were implemented in the clinical setting in the two UNESCO biosphere areas. Two additional arms were offered as a preventive programme for the general population in the UNESCO Biosphere Reserve Berchtesgadener Land. A schematic representation can be found in Figure 2.



Figure 2. Design of the Green Care study

Arm 1

Contrary to the original plan, different interventions were offered at both sites. What they have in common is that they are based on guided contact with nature and emphasise experience in nature as a therapeutic factor. The content of the interventions was the responsibility of the respective UNESCO biosphere area staff.

In implementation, somewhat different focal points were set: The procedures used in the UNESCO Biosphere Reserve Rhön were relaxation-based ("nature-based relaxation offering"), while they were mindfulness-based ("nature-based mindfulness training") in the UNESCO Biosphere Reserve Berchtesgadener Land. These two terms are used in the site-specific evaluations. In the remainder of the final report, the term "Green Care" is used to describe the mindfulness- or relaxation-based interventions grounded in nature, in line with the scientific research.

The intervention offerings were designed for small groups with a maximum of 6 participants and were delivered once a week for 3 to 4 hours over a period of 3 weeks (UNESCO Biosphere Reserve Berchtesgadener Land) or 4 weeks (UNESCO Biosphere Reserve Rhön).

While the offering in the UNESCO Biosphere Reserve Rhön was carried out in closed groups, the intervention in the UNESCO Biosphere Reserve Berchtesgadener Land included ongoing groups. These structural differences were due to the different conditions of the cooperating clinics.

A detailed description of the respective methods can be found in the final report of the intervention team.

Arm 2

The second clinical arm corresponded to Treatment As Usual plus Waiting List (TAU+WL) with a shortened offering after the second data collection in the form of a Waiting List control. Although the offering for this group was significantly reduced, it is still appropriate to refer to it as a Waiting List because patients were informed of the offering at the beginning of the study and received it specifically as compensation for their willingness to participate in the study.

Arm 3

In the UNESCO Biosphere Reserve Berchtesgadener Land, the Green Care treatment programme ("nature-based mindfulness training") was also offered to the general population to the same extent as in Arm 1.

Arm 4

The fourth preventive arm was designed as a Waiting List control group (WL). Here, participants received the questionnaires at T1 and T2 and then a short offering after the second data collection.

2.6 Evaluation methods

2.6.1 Questionnaire

Statistical analyses were performed using SPSS Statistics 29 (IBM Corp., Armonk, NY, USA) and R (R Core Team, 2020).

Analyses were performed separately for the clinical and preventive arms. Linear mixed models with repeated measures (MMRM) were used to compare the trajectories of the dependent variables based on the factors time (T1 and T2) and group (Green Care and TAU+WL or Green Care and WL). In addition, the site (UNESCO Biosphere Reserve Rhön and UNESCO Biosphere Reserve Berchtesgadener Land) was used as a factor to control for potential differences between the two sites.

As follow-up data were only available for the Green Care groups, additional MMRM analyses were conducted with the dependent variables using the Green Care groups and the time factor (T1, T2 and T3), again controlling for site.

All MMRM analyses used were analysed with fixed effects and random intercepts. Model estimation was based on full information maximum likelihood. The significance level is set at p < .05 for all calculations. Trends are indicated when results reach a significance level of p < .10 and are associated

with a hypothesis. This approach can provide directions for further research while mitigating the problem of post-hoc interpretation.

To balance the two groups (Green Care and TAU+WL or Green Care and WL) and to compensate for the shortcomings of a non-randomised sample, a propensity score was calculated for each participant. This score was defined as the predicted probability of choosing the Green Care treatment using available sociodemographic and psychometric assessments through logistical regression analyses. Each participant received a score ranging from 0 (control group) to 1 (Green Care group). This score was then used as a covariate in the MMRM analyses.

In further sensitivity analyses (cf. chapters 3.4 and 4.3), variables such as depression (only in the clinical sample), gender, urban vs. rural environment in adolescence, and previous experience with similar training were included in subgroup analyses.

2.6.2 Process evaluation

Data from the process evaluation regarding participants' subjective well-being during the intervention and the subjective effectiveness of the intervention were analysed using Microsoft Excel (Microsoft Corporation). Means were calculated and the standard deviation was used as a measure of scattering.

In the clinical study in the UNESCO Biosphere Reserve Rhön and in the prevention trial, both variables could be analysed over time for each session group: well-being and effectiveness in the first, second and third session and, in the UNESCO Biosphere Reserve Rhön, in the fourth. This was not possible in the clinical study in the UNESCO Biosphere Reserve Berchtesgadener Land due to the continuous intervention groups. Here, the average well-being and effectiveness from the perspective of the intervention participants were calculated across all session dates.

SPSS Statistics 29 (IBM Corp., Armonk, NY, USA) and R (R Core Team, 2020) were used to integrate the process evaluation into the subgroup analyses. Relevant variables included session group size, weather conditions (sunny vs. cloudy), and temperature during the session.

Other situational aspects were used to provide a more detailed description of the intervention setting.

2.6.3 Interviews

The interview data from the clinical trial and the prevention trial were analysed separately. The audio material was transcribed according to the recommendations of Flick (2017) and Dresing and Pehl (2015). The written material was then evaluated with a summarising content analysis involving inductive category formation according to Mayring (2016). The content of the interviews was reduced in such a way that all essential information was retained, resulting in a corpus (category system) that clearly represented the source material. This category system was supplemented with definitions and anchor examples for each category, an approach that is part of scaling structuring within structured content analysis (Mayring, 2016). Coding rules were established for categories that were difficult to distinguish from each other.

In a first step, two randomly selected interviews for the clinical and prevention trials were analysed in this way, creating at least one category system for each construct under investigation. The material

from the remaining interviews was then integrated into these category systems. Additional categories were created and existing categories were adapted as necessary. To enable a quantitative analysis of the interview data, the frequency of mentions for each category created and the number of experts who expressed an opinion about it were determined in absolute and percentage terms. Finally, the categories of a category system were arranged in order of frequency of mention.

The postscripts of the interviews were used to check whether any bias in the data could be expected due to situational aspects or topics of conversation that occurred before or after the recording. This was not the case, so all interviews could be included in the data analysis.

2.7 Study quality

The quality of quantitative studies is assessed on the basis of the three central quality criteria of test theory: objectivity, reliability and validity (Döring & Bortz, 2016).

- Objectivity of the study: Through the use of standardised questionnaires, the results of the study are independent of the person conducting the investigation, which indicates *objectivity in the implementation*. In addition, the data collected yield the same results when analysed by different evaluators, thus ensuring the *objectivity of the evaluation*. Finally, the evaluation of a statistical value is not influenced by the interpretation of the evaluator, thus ensuring the *objectivity of the interpretation*.
- Reliability of the study: The accuracy and precision of the measurement is confirmed using Cronbach's Alpha.
- Validity of the study: As the scales and items have been formulated and operationalised on the basis of theory and, where possible, existing scales have been used, there is an adequate level of *content validity*. *Construct validity* is confirmed by factor analyses. Ecological validity is demonstrated by the similarity of the study setting to the natural environment and the transferability of the study findings to everyday life and practice.

The quality of the qualitative study is assessed on the basis of Steinke's (1999) seven criteria for evaluating qualitative research: (1) the data collection and analysis procedure is intersubjectively comprehensible, (2) there is an indication of the use of a qualitative research approach, (3) the study is empirically grounded, (4) limitations of the study are reported, (5) the subjectivity of qualitative research is reflected, (6) coherence is considered, and (7) the relevance of the study topic is reported.

3. Clinical trial results

3.1 Description of the sample

A total of 251 patients were invited to participate in the trial at both centres (cf. Fig. 3). Of these, 21 (8 per cent) declined to participate and three individuals did not meet the inclusion criteria. There were 116 patients allocated to the Green Care group and 111 patients allocated to the TAU+WL group. These patients will be included in the intent-to-treat analysis in future assessments, even if incomplete data are available. In the Green Care group, 13 people did not receive the intervention, resulting in a dropout rate of 11 per cent. In the TAU+WL group, 8 patients (7 per cent) dropped out.



Figure 3. Clinical trial flow chart

In the end, 103 patients in both the Green Care and TAU+WL groups received the intervention. Completed questionnaires at T2 are also available for these patients.

For the follow-up three months after the intervention, only the 103 participants in the Green Care group who completed the training were contacted. Of these, 92 returned completed questionnaires, resulting in a response rate of 89 per cent.

Demographic and clinical data at baseline T1 are shown in Table 1.

Characteristics of the clinical sample at baseline				
	Green Care group (n = 116)	TAU+WL group (n = 111)		
Gender	61.2% f, 38.8% m	55.0% f, 44.1% m, 0.9% unknown		
Age	<i>M</i> = 52.25; <i>SD</i> = 8.51; MIN = 29; MAX = 65	<i>M</i> = 50.95; <i>SD</i> = 10.45; MIN = 19; MAX = 64		
Children	69.0% yes, 31.0% no	66.7% yes, 32.4% no		
Training	17.2% yes, 81.0% no, 1.7% unknown	20.7% yes, 78.4% no, 0.9% unknown		
Positive affect	<i>M</i> = 2.46, <i>SE</i> = 0.08	<i>M</i> = 2.50, <i>SE</i> = 0.08		
Negative affect	<i>M</i> = 2.98, <i>SE</i> = 0.08	<i>M</i> = 2.88, <i>SE</i> = 0.08		
Depression	<i>M</i> = 12.32, <i>SE</i> = 0.50	<i>M</i> = 12.31, <i>SE</i> = 0.51		

Table 1

Experts for the interviews

Ten experts in the clinical setting participated in the interviews for the qualitative evaluation of the intervention (70 per cent female). The professional backgrounds of the interviewees were diverse, with some having multiple qualifications: The experts interviewed included the two project facilitators, two chief physicians, three (neuro)psychologists, a reference therapist, a nurse, a psychotherapist, two mindfulness trainers and a ranger. The age distribution of the experts is shown in Table 2. On average, the respondents had frequent contact with patients in the clinic (M = 3.00; SD = 0.77; Table 3) and were well informed about the intervention (M = 3.20; SD = 0.87; Table 4). According to the experts, the intervention is highly effective (M = 3.70; SD = 0.46; Table 5).

Table 2

Age distribution of experts in the clinical trial					
	< 30	30-40	40-50	50-60	> 60
N	2	2	2	4	0
Note. A	ge in years				

Table 4

Contact of experts with patients in the clinical trial

	1	2	3	4
N	0	3	4	3
Note. 1	= I have al	most no con	tact with pa	itients. 4 = I

have a lot of contact with patients. Table 3

Clinical trial experts' knowledge of intervention

	1	2	3	4	
Ν	0	3	2	5	

Note. 1 = I know absolutely nothing about it. 4 = I am familiar with the offering.

Table 5

Overall evaluation of the offering by experts in the clinical trial

	1	2	3	4
N	0	0	3	7

Note. 1 = I do not find it very effective. 4 = I find it very effective.

3.2 Main results

3.2.1 Questionnaire

Mixed models with repeated measures (MMRM) were used for the main data analyses, i.e., the group comparisons between the Green Care and TAU+WL groups at T1 and T2. Time point, group and site, along with their interactions, were included in the analyses as fixed factors. A random intercept was also included in the analyses. The propensity score was used as a covariate to account for differences between groups due to non-randomised group allocation. According to the research questions, it was expected that there would be significant interactions between time point and group, but not between time point, group and site.

Primary outcomes: Positive and negative affect

When comparing the two groups at time points T1 and T2, significant Time*Group effects were observed for the two PANAS subscales, Positive affect and Negative affect (Positive affect: F = 7.15, df = 1/207.15, p = .008; Negative affect: F = 10.52, df = 1/211.83, p = .001). The interaction of Time*Group*Site was not significant for either variable. Figure 4 shows the estimates of the scores from the analyses. The effect sizes indicate that the Green Care group had strong effects (Positive affect: d = 1.13, Negative affect: d = 1.04), while the TAU+WL group also showed significant effects, but with medium effect sizes (Positive affect: d = 0.76, Negative affect: d = 0.59).



Figure 4. Positive and negative affect in the clinical trial: comparing the Green Care and TAU+WL groups from T1 (admission) to T2 (discharge)

Looking at the longitudinal analysis of the Green Care group (Fig. 5), a significant effect was also observed after three months (T3) compared to the baseline (T1). However, the effects decreased slightly compared to the second time point immediately after treatment (decrease in positive affect: small effect with d = 0.23, decrease in negative affect: small effect with d = 0.21).



Figure 5. Positive and negative affect in the Green Care group of the clinical trial (T1-T2-T3)

Secondary outcomes

Significant interactions were found for the secondary outcomes of self-compassion and emotional connectedness to nature. For both variables, significantly larger effects were observed in the Green Care group compared to the TAU+WL group. Medium effects also remained at the three-month follow-up.

Non-significant interactions were found for depression, mindfulness, social support, nature-related mindfulness, awareness of hazards, attribution of responsibility, nature conservation attitudes and behaviour, and current contact with nature. However, there are clear trends towards a larger effect size for mindfulness (p = .098) in the Green Care intervention. These trends would need to be further investigated in a study with greater statistical power. It is worth noting the equality of effect sizes for current contact with nature. Green Care and TAU+WL patients did not differ in the frequency of contact with nature at T2. Thus, the difference in experiences in nature seems to be related to the quality of the experiences in nature, not the frequency of the time spent in nature. In the Green Care group this is characterised by guided contact with nature.

Detailed tables and figures with the results of the primary and secondary outcomes can be found in Annex C.1.

3.2.2 Process evaluation

Based on the data from the process evaluation, it becomes clear that the Green Care interventions in the clinical trial were carried out under very different conditions. Nevertheless, (highly) significant effects can be experimentally demonstrated, indicating the high efficacy of nature-based interventions in a clinical setting.

Overall, the sessions took place throughout the year, although in the UNESCO Biosphere Reserve Berchtesgadener Land the period was limited to June to November due to weather conditions. The

interventions took place at different times (between 8:00 am and 5:30 pm). The average group size was 6 people, with a range from 2 to 12 participants. The interventions were conducted at temperatures ranging from -13°C to 32°C, and weather conditions included sunshine, fog, clouds, wind, storms, rain showers, thunderstorms and snowfall.

96 per cent of the patients in the Green Care group reported feeling comfortable during the sessions (M = 3.68; SD = 0.55). Across all sessions, 94 per cent of the patients rated the interventions as effective (M = 3.55; SD = 0.63). The experimentally demonstrated effectiveness, as evidenced by quantitative data from questionnaires and expert ratings of the effectiveness of the interventions in interviews, is also confirmed from the patients' perspective by the process evaluations.

3.2.3 Interviews

The main findings from the interviews with experts in the clinical trial, which focused primarily on the effectiveness of Green Care, are provided below. To start with, the project characteristics considered relevant to success are presented, followed by the effectiveness of the intervention on biopsychosocial health and environmental behaviour. The factors influencing the individual and intervention levels, as well as the influence of the intervention site (nature), conclude the presentation of the interview results. Complete category systems with anchor examples, frequency of experts who made statements on the category and, where necessary, coding rules for the presented primary and secondary constructs can be found in Annex B.1.

Success-relevant project characteristics

The experts have a nuanced understanding of the factors that influence whether the intervention can be implemented successfully (see Fig. 6).



Figure 6. Success-relevant project characteristics of a nature-based intervention in the clinical setting

Good *cooperation* between project staff and clinics is seen as the most important prerequisite for successful implementation of the intervention (e.g., E8: "The activity facilitator and I [clinic staff], we give each other space to work well together. Tolerance, scheduling, agreements, etc. work quite well"). The effective *exchange of information* between the participants is also considered important, which includes both the documentation of relevant information (e.g., E6: "In the documents I received, I found that what is wanted, what it's about, is very transparent. I found the information sufficient, but not overwhelming.") and the early notification of the clinic staff about the upcoming intervention (e.g., E3: "Of course, it took some time for it to trickle down, so to speak, to actually inform the doctors and the treating psychotherapists").

Considered relevant for success are also an appealing and appropriate *naming of the offering* (e.g., E3: "The first thing I tell them is not to call it forest bathing."), *flexibility* in the implementation of the intervention depending on the participants and situational conditions (e.g., E7: "We always adapt. We talk again the evening before the intervention, and after the intervention, after the lunch break, we check it again, and it is always adjusted."), as well as an organised and structured *recruitment* of the participants (e.g., E5: "Only patients who meet the inclusion criteria and are eligible for the project are informed during the recruitment week that they are suitable for this project and whether they can imagine participating in it.").

Finally, the aspect of *coordinating the timing* of the intervention with other clinic offerings or therapies is mentioned (e.g., E13: "Originally, the offering was four hours long. This was difficult because, for example, the meal times in the clinic were extended because of corona ... and so the time had to be shortened a little"). These organisational factors are complemented by the choice of a *suitable intervention site* for implementing the offering. This site must be easily accessible to the patients and provide a suitable environment for the nature-based intervention.
Effects on mental well-being

In total, the experts mention 16 different categories of mental well-being in which they observe changes due to the Green Care intervention (see Fig. 7).



Figure 7. Effects of the intervention on mental well-being in the clinical setting

In particular, according to the experts, Green Care induces *mental relaxation and calmness* (e.g., E5: "This is very beneficial for the patients; somehow it relaxes them."), builds *positive emotions* (e.g., E8: "Mood improves."), as well as a *directing and focusing of perception* (e.g., E3: "... this altered perception, and indeed also this systematic attention guidance"). Furthermore, the intervention is seen as a measure through which patients can *distance themselves from their everyday lives* (e.g., E5: "... where you really notice how the patients step out of their usual daily routines or even the clinic routine here").

In addition to these effects, there are a number of other effects of the intervention that are also perceived by the experts, albeit less frequently. These pertain to the areas of *stress management* (e.g., E3: "... in order to possibly avoid stressors or engage in a form of stress management.") and *concentration* (e.g., E2: "What patients reported, was, for example ... concentration."), *self-care* (e.g., E2: "Self-care was also promoted."), *cognition* (e.g., E3: "I firmly believe that this offering also has effects on cognition.") and *reflection* (e.g., E10: "And a lot of it came naturally through reflection."). Effects on *self-perception, frustration tolerance, cognitive flexibility, sense of freedom and mindfulness* are reported by one respondent each.

Concerning the applicability of the intervention to patients with mental disorders, the suitability for patients with *depression* is mainly mentioned (e.g., E3: "...an offering that is primarily designed for affective disorders or depressive disorders. Therefore this sample benefited the most from the

offering"). Furthermore, there are other mental illnesses for which the intervention is perceived as suitable and effective (see Fig. 8).



Figure 8. Applicability of the intervention to diverse disorders in the clinical setting

This applies firstly, in general terms, to patients with *different mental disorders* (e.g., E3: "In general, I actually think that the offering would be suitable for almost all mental disorders …"). In particular, mention is made of *obsessive perfectionism* (e.g., E3: "We've had a handful of people so far who, at least from my perspective, have had some difficulties with the pursuit of perfectionism … and especially with this symbolic work I've had very good results with these specific patients."), *acute stress reactions, chronic fatigue syndrome*, and *schizophrenia*.

Effects on physical well-being

The experts identified eight areas of physical well-being that were affected by Green Care (see Fig. 9).





The intervention is perceived as motivation to *increase physical activity*, which has an impact during the sessions (e.g., E3: "... it's not about getting from A to B quickly, but rather everyone goes at their own pace ...It can take as long as necessary for individuals, and there's no performance pressure.") and beyond (e.g., E13: "Some patients saw it as a motivation to do this more often because they noticed that it was physically beneficial for them").

At the same time, a number of other effects were observed: a *reduction in physical tension and restlessness* (e.g., E2: "On the physical side, four hours in the forest, i.e. in a low-stimulation environment, contributed a lot to relaxation"), *deepening of breathing* (e.g., E3: "Breathing is also

often reported by the patients as calmer and more controlled, also more balanced."), as well as an *improvement in sleep quality* (e.g., E13: "Patients often reported that they slept better after the offering; many have sleep disorders, and it was better right after the offering; they could sleep better at night.") and *body awareness* (e.g., E2: "Body awareness ... was also promoted.").

Similarly, the experts have observed effects of the intervention concerning the *strengthening of the cardiovascular system* (e.g., E13: "... because they noticed that it benefits them physically, that it strengthens their circulation"), a *reduction in headaches* (e.g., E3: "One participant said that she no longer had headaches") and an *improvement in posture*.

Effects on social well-being

In terms of social well-being, the experts mentioned five effects of the intervention in their interviews (see Fig. 10).





During the sessions, the experts observe an *increasing openness* (e.g., E2: "They are also more open and willing to talk, even ready for intimate talks.") and *mutual consideration* among the patients (e.g., E5: "... then they also look at each other very attentively and respectfully."). In their opinion, *a sense of belonging to the group* was promoted (e.g., E8: "They know each other from the project and then, in the clinic during meals, or anywhere else, they have a different connection to each other than those who did not participate in the project") and *trust among the patients was built up* (e.g., E2: "Therefore, patients who are more reserved were more relaxed, and patients who are anxious were also calmer"). In general, *social skills are promoted* (e.g., E3: "We have an element that we do every time for four weeks, which is definitely meant to strengthen social skills").

Effects on pro-environmental and nature conservation behaviour

The experts' statements indicate that the intervention influenced four pro-environmental and nature conservation behaviour (see Fig. 11).



Figure 11. Effects of the intervention on pro-environmental and nature conservation behaviour in a clinical setting

The central effect of the intervention on pro-environmental and nature conservation behaviour is the observation of a *more mindful perception of nature* (e.g., E5: "Then, in the next step, one becomes more aware that this is also worth protecting"). Furthermore, there is an encouragement for *engagement in ecological behaviour* (e.g., E7: "...everyone is thinking about the trash ... and everyone notices that it's increasing, and there are discussions about how to change that.") and patients also pay more attention to *reducing litter in nature* (e.g., E2: "...bottles were found, and the participants removed them voluntarily and disposed of them"). Finally, patients are observed as being more *mindful in their interactions with nature and animals* (e.g., E10: "... during the activities in the forest, care was taken to ensure that nothing was damaged or torn down").

Overall effectiveness

A category system can also be created for the experts' assessment of overall effectiveness (see Fig. 12).





In the interviews, there are several statements that indicate that the intervention is perceived as *very effective* (e.g. E3: "...overall I believe that this is a very effective and beneficial thing."). A difference is specifically made between *short-term* (e.g., E8: "...already after the first time, the patients started to benefit from it."), *medium-term* (e.g., E2: "I would say ... a medium-term effect.") and *long-term effectiveness* (e.g., E5: "That could also be a long-term effect.").

Individual determinants of effectiveness

The experts mention several factors that influence the reported effectiveness. At the patient level, nine such determinants can be identified (see Fig. 13).





Furthermore, *individual factors* in general (e.g., E2: "I can imagine that it depends on the person."), the *physical condition* of the patients (e.g., E13: "There were also individual patients whose physical condition was not so good for the offering. For example, patients who were very overweight and found the walk too long. Also, patients with post-Covid who had major lung problems."), *pre-existing stressors* (e.g., E5: "So, how acutely stressed the person is at the moment, because it can be harder to set aside current issues and engage in the offering."), and *previous experiences* (e.g., E3: "And I have the feeling that for those who are relatively new to it, it could possibly have a greater effectiveness because they are new to this nature theme and may be more fascinated by it.") were identified as determinants of the effectiveness of the offering.

Finally, factors influencing the effectiveness of the intervention include the *receptiveness of the patients for the offering* (e.g., E8: "... whether the patients are receptive or not."), *subsequent daily stress* (e.g., E2: "... after these projects, I had to return to the daily stress right after lunch, and the effect was not particularly significant."), and *gender* (e.g., E13: "One difference I noticed between men and women: men often expressed that they experience a sense of freedom in the forest.").

Intervention-related determinants of effectiveness

At the intervention level, the experts report four different parameters that influence the effectiveness of Green Care (see Fig. 14).



Figure 14. Determinants of session-level effectiveness of the intervention in the clinical setting

From the experts' point of view, the *scope of the offering* is crucial for the effectiveness of the intervention. They believe that a minimum number of sessions should be required (e.g., E3: "I think that a minimum of four sessions should be required. I think that three or two are not enough.") and that a maximum number of sessions should not be exceeded (e.g., E3: "... therefore I think that doing it once a week is a good plan"). The duration of each session also has an impact on effectiveness (e.g., E7: "If it's less than four hours, then it's useful"). A *minimum of two-time participation* in the intervention is recognised as necessary (e.g., E3: "... in the second week, often right at the beginning, even before it starts, the patients already come with a completely different mindset").

On the group level, the *composition of the groups* (e.g., E3 "... that has a lot to do with the composition of the groups, whether such a dynamic develops.") and *group size* (e.g., E2: "The group was quite small, so it was a bit more intense") are perceived as factors contributing to the effectiveness of the intervention.

Effectiveness due to the natural setting

In addition to these individual and organisational factors, the experts mention aspects arising from the setting of the intervention in nature and its influence on effectiveness (see Fig. 15).



Figure 15. Determinants of intervention effectiveness due to the natural environment in the clinical setting

Nature as the site at which the intervention takes place is considered to be of great importance and special *significance* (e.g., E3: "But ultimately this natural environment is decisive; it provides nature; it provides space"). Spending time in nature awakens *memories* of the past (e.g., E13: "... that there are links to childhood, for example"). The intervention is also effective because there is an innate *connectedness to nature* (e.g., E8: "I am convinced that we all have these aspects within us, this connection to nature. These aspects are more or less hidden in us human beings").

The specific *design* of the natural environment is perceived as less crucial (e.g., E5: "If we were to do the whole thing 500 metres further away, I assume it would work just as well"). More relevant for effectiveness are *sensory experiences* in nature (e.g., E7: "These are different sounds, it smells different, it looks different every day."), the *naturalness* of the intervention site (e.g., E5: "It's about the fact that this is nature, where nature is left as it is.") and *being outdoors* in itself (e.g., E3: "... it can never be as good as when you see it on site. That's why I think it's very different from being in a room").

According to the experts, this is because nature can serve as an intimate *retreat* (e.g., E3: "... so it was something personal for her that she shared with the others, this place."), which is *quiet* (e.g., E1: "... to enjoy this peace even more in the forest.") and *low in stimuli* (e.g., E10: "I think the significance of it was that there was just no other habitat present. It was just the forest. For me it was a kind of sensory deprivation, low in stimuli"). At the same time, nature harbours great *diversity* (e.g., E7: "... that it can be so different, can look different.") and *energy* (e.g., E1: "This place is characterised by a special kind of energy and, in my view, also has an effect on the human psyche, it is purely energetic.") and enables

the experience of *vastness* (e.g., E3: "... is not closed at the top like an indoor space, but because the tall trees form this roof, so to speak").

Furthermore, the elements of the intervention have a *strong connection to nature* (e.g., E3: "There are also approaches to nature-based interventions that take place indoors in bad weather, for example, that would not work at all with our offering, that would do nothing."); nature can have an *aesthetic* effect (e.g., E7: "...when we are sitting in the 'beautiful' forest.") and is in principle *freely accessible* (e.g., E5: "Anyone can be in nature, and it doesn't require special skills or knowledge."), which favours its effectiveness.

3.3 Clinical trial results by site

In order to better understand the effects of the interventions in the two UNESCO biosphere areas, the results will be discussed separately for each site. First, it should be noted that there were no significant differences between the sites in the effectiveness of the Green Care interventions on the primary outcome measures. This means that the effectiveness of the intervention was demonstrated for both sites. When analysed separately, there is less statistical power to detect smaller effects, which means that individual differences may not always be statistically significant. Nevertheless, separate analysis can provide a more detailed picture.

In this section, analyses have been carried out separately for both sites. The focus of the discussion is on the primary outcome measures (positive and negative affect), but the secondary outcome measures are addressed briefly. Detailed results can be found in Annex C.1.

3.3.1 UNESCO Biosphere Reserve Rhön

Primary outcomes: Positive and negative affect

Group comparison between T1 and T2

In the UNESCO Biosphere Reserve Rhön, there was a significant Time*Group interaction for the primary outcome measure of Positive affect (F = 7.28, df = 1/98.34, p = .008). Positive affect increased significantly in the "Nature-based relaxation offering" group with a very large effect ($d_{Nature-based relaxation}$ offering = 1.44) and in the TAU+WL group with a large effect ($d_{TAU+WL} = 0.91$). For the primary outcome measure of Negative affect, the Time*Group interaction was not significant (F = 3.67, df = 1/105.64, p = .058). However, there was a trend confirming that the effect was greater in the "Nature-based relaxation offering" group than in the TAU+WL group ($d_{Nature-based relaxation offering = 1.21$ vs. $d_{TAU+WL} = 0.83$; see Fig. 16).



Figure 16. Positive and negative affect in the clinical study (UNESCO Biosphere Reserve Rhön): comparison between "Nature-based relaxation offering" group and TAU+WL group from T1 (admission) to T2 (discharge)

Follow-up in the "Nature-based relaxation offering" group

For Positive affect variable, a large effect remained in the comparison of T1 to T3 ($d_{T1T3} = 0.88$). The decrease from T2 to T3 was not significant (p = .177). For Negative affect, a medium effect remained in the comparison of T1 to T3 ($d_{T1T3} = 0.64$, see Fig. 17). The decrease from T2 to T3 was significant (p = .040).



Figure 17. Affect T1-T2-T3 in the clinical trial (UNESCO Biosphere Reserve Rhön)

Process evaluation

The closed groups allowed for an in-depth analysis of the process evaluation to be conducted, focusing on subjective well-being and subjective effectiveness on a session-specific basis. It is found that patients felt very comfortable during the intervention from the first session onwards (M = 3.51; SD = 0.38), and this perception continued to increase until the last session (M = 3.94; SD = 0.10). A similar trend was observed for the effectiveness of the intervention from the patients' perspective: The offering was already perceived as effective after the first session (M = 3.37; SD = 0.40), and even more so after the fourth and final session (M = 3.85; SD = 0.22; see Fig. 18).



Figure 18. Subjective well-being and subjective effectiveness of the intervention over time from the perspective of the clinical trial participants.

(Data from UNESCO Biosphere Reserve Rhön; 1 = I did not feel comfortable; 4 = I felt very comfortable; 1 = I did not find it effective; 4 = I found it very effective.)

Process evaluations were also carried out in the control groups. These evaluations showed that patients in the shortened interventions, consisting of two sessions, felt very comfortable overall (M = 3.91; SD = 0.19) and perceived the sessions to be effective (M = 3.78; SD = 0.27). Subjective well-being was already high in the first session, with a minimal further increase (1st session: M = 3.90; SD = 0.22; 2nd session: M = 3.93; SD = 0.13). Regarding the effectiveness of the nature-based relaxation offering, there was a significant increase between the first and second session (1st session: M = 3.69; SD = 0.29; 2nd session: M = 3.93; SD = 0.13).

3.3.2 UNESCO Biosphere Reserve Berchtesgadener Land

Primary outcomes: Positive and negative affect

Group comparison between T1 and T2

In the UNESCO Biosphere Reserve Berchtesgadener Land, there was no significant Time*Group interaction for the primary outcome measure of Positive affect (F = 1.22, df = 1/107.52, p = .271). However, it was observed that Positive affect increased significantly in the "Nature-based mindfulness training" group with a large effect ($d_{\text{Nature-based mindfulness training}} = 0.83$) and in the TAU+WL group with a medium effect ($d_{\text{TAU+WL}} = 0.61$). For the primary outcome measure of Negative affect, the Time*Group interaction was significant (F = 7.10, df = 1/106.09, p = .009). In this case, the "Nature-based mindfulness training" group showed a large effect, whereas the TAU+WL group showed a small effect ($d_{\text{Nature-based mindfulness training}} = 0.87$ vs. $d_{\text{TAU+WL}} = 0.35$; see Fig. 19).



Figure 19. Positive and negative affect in the clinical trial (UNESCO Biosphere Reserve Berchtesgadener Land): comparison between "Nature-based mindfulness training" group and TAU+WL group from T1 (admission) to T2 (discharge)

Follow-up in the "Nature-based mindfulness training" group

For Positive affect variable, a small effect remained in the comparison of T1 to T3 (d_{T1T3} = 0.40). The decrease from T2 to T3 was significant (p = .030; see Fig. 20). For Negative affect, a medium effect remained in the comparison of T1 to T3 (d_{T1T3} = 0.48). The decrease from T2 to T3 was not significant



(p = .259).



3.4 Subgroup analyses

To assess the validity of the results for different subgroups, it is necessary to consider them in subgroup analyses. Methodologically, this presents the challenge of reduced statistical power to detect effects and the risk of accumulating errors. The latter can lead to non-existent effects becoming statistically significant by chance. To avoid this error, two measures were taken: firstly, the analyses were restricted to the primary outcome variables of the PANAS, and secondly, the accepted significance level was corrected to p = .01. This is indicated in the discussions of effects below.

Depression

The PHQ-9 scale was divided based on the cut-off score of 10, following the recommendations of Meanea et al. (2012). Individuals with a score of less than 10 were categorised as "no or mild depression", whereas those with a score of 10 or more were categorised as "possible major depression". The resulting binary variable was included in the analyses.

In the group comparison of Green Care vs. TAU+WL, there were no significant interactions for the primary outcome variables (Positive and Negative affect): Patients with low and high levels of depression benefited equally from the interventions.

In the longitudinal analysis, considering only the Green Care group, significant interactions were found for the primary outcome measures (Positive affect: p < .001; Negative affect: p < .001). The results are a product of two circumstances: Firstly, at T1, more highly distressed patients had significantly less favourable values in clinically relevant variables, leading to an even more pronounced improvement. Secondly, the results suggest that the effects are more stable in more depressed patients, as the values between T2 and T3 do not significantly differ for this group, whereas in the group of patients with lower levels of depression, these values slightly decrease again or, in the case of Negative affect, slightly increase. In summary, it can be said that more highly distressed patients particularly benefited from the program.

Childhood residence

The Childhood residence variable was used as a binary variable (urban/rural) in the analyses. In a group comparison of Green Care vs. TAU+WL, the results showed no significant difference for the primary outcomes. This means that the origin of the participants had no direct influence on the effectiveness of the intervention.

In the longitudinal analysis, including the follow-up measurement time point, there was a significant interaction for Negative affect: Here, more stable effects were found at T3 for people from an urban environment. This means that the values for this group did not differ significantly between the time of discharge and three months after treatment.

Gender

The Gender variable was asked in three categories in the study (female, diverse, male). As none of the people in the questionnaire classified themselves as "diverse", this category is not considered further in the analyses. The results also show for the gender variable that no significant differences were found in the primary outcome measures.

Previous experience

The "Previous experience" variable asked about previous experience with mindfulness training or training in nature. There was no significant effect for the primary outcome measures, either in the T1-T2 group comparison or in the longitudinal analysis for the Green Care group.

Age

Age was included as a binary variable by using a median split (median = 55 years). No significant results were found for the primary outcome measures, either in the comparison of Green Care vs. TAU+WL or in the longitudinal analysis for the Green Care group.

Children

In the questionnaire, participants were asked whether they had children, with the response options yes and no. This variable showed no significant effects on the primary outcome measures, both in the comparison of Green Care vs. TAU+WL or in the longitudinal analysis for the Green Care group.

Weather

The influence of weather was examined using process evaluation data. Two factors were taken into account: firstly, whether the sun was shining predominantly during the intervention (dichotomous variable), and secondly, the temperature during the intervention (the highest temperature during the intervention was chosen in each case). These variables were then correlated with the results of the process evaluation (well-being and effectiveness). Only very small and non-significant correlations were found ($r_{\text{Sunshine well-being}} = .11$, p = .41; $r_{\text{Sunshine effectiveness}} = .08$, p = .55; $r_{\text{Temperature well-being}} = -.13$, p = .37; $r_{\text{Temperature effectiveness}} = -.06$, p = .68).

4 Results of the prevention trial

The prevention trial was carried out in the UNESCO Biosphere Reserve Berchtesgadener Land. Therefore, the evaluations refer only to this site.

4.1 Description of the sample

A total of *N* = 146 people expressed interest in the study (cf. Fig. 21). Of these, 13 individuals (5 per cent) were not included in the study due to inclusion criteria (8) or other reasons (5). In the end, 84 people were assigned to the "Nature-based mindfulness training" group and 49 people to the WL group. These individuals are included in the intent-to-treat analysis in further assessment, even if incomplete data were available. In the "Nature-based mindfulness training" group, 28 people ultimately did not receive the intervention. Of these, 17 people (20 per cent) were unable to participate in the programme because of a lockdown imposed to contain the SARS-COV2 virus. A further 11 people (13 per cent) were unable to take part in the programme at short notice because of their health condition. There were no premature withdrawals from the trial in the WL group.



Figure 21. Flow chart of the prevention trial

In the end, 56 people in the "Nature-based mindfulness training" group and 49 people in the WL group received the intervention. Fully completed questionnaires were available for these individuals at T2.

For the follow-up three months after the intervention, only the 56 participants in the "Nature-based mindfulness training" group who completed the training were contacted. Of these, 47 returned completed questionnaires, resulting in a response rate of 84 per cent.

Demographic and relevant psychometric data at baseline (T1) are presented in Table 6.

Table 6 Characteristics of the prevention sample at baseline

	"Nature-based mindfulness training" group (n = 67)	WL group (n = 49)
Gender	86.6% f, 13.4% m	71.4% f, 28.6% m
Age	<i>M</i> = 48.72; <i>SD</i> = 11.22; MIN = 24; MAX = 68	<i>M</i> = 42.76; <i>SD</i> = 17.20; MIN = 18; MAX = 85
Children	74.6% yes, 25.4% no	51.0% yes, 49.0% no
Training	40.3% yes, 59.7% no	32.7% yes, 67.3% no
Positive affect	<i>M</i> = 3.07, <i>SE</i> = 0.10	<i>M</i> = 3.06, <i>SE</i> = 0.11
Negative affect	<i>M</i> = 2.47, <i>SE</i> = 0.09	<i>M</i> = 2.30, <i>SE</i> = 0.11

Green Care - Final Report of the Scientific Monitoring Study

Experts for the interviews

Seven experts (71 per cent female) took part in the interviews as part of the prevention trial. People with different roles in the intervention and different professional qualifications were recruited for the interviews, including the project facilitator, a landscape planner, a nature coach, a mental psychotherapist and a health educator. The age distribution of the experts is shown in Table 7. The interviewees are well informed about the intervention (M = 3.43, SD = 0.73; Table 8) and subjectively perceive it as highly effective (M = 3.71; SD = 0.45; Table 9).

Table 7 Age distribution of experts in the prevention trial

	< 30	30-40	40-50	50-60	> 60
N	1	1	4	0	1

Note. Age in years

Table 9

Overall evaluation of the offering by experts in the prevention trial

	1	2	3	4
N	0	0	2	5

Note. 1 = I do not find it very effective. 4 = I find it very effective.

Table 8

Prevention trial experts' knowledge of intervention

	1	2	3	4
N	0	1	2	4

Note. 1 = I know absolutely nothing about it. 4 = I am familiar with the offering.

4.2 Main results

4.2.1 Questionnaire

Mixed models with repeated measures (MMRM) were used for the main data analyses. Fixed factors included time point of measurement, group and their interactions in the analyses. A random intercept was also included in the analyses. The corresponding propensity score was used as a covariate to adjust for differences between the preventive groups due to non-randomised group assignment. According to the research questions, significant interactions between time point and group were expected. It should be noted that the slightly smaller sample size compared to the clinical arms is expected to result in lower statistical power.

Primary outcomes: Positive and negative affect

When comparing the two groups at time points T1 and T2, there was no significant interaction between time point and group for the Positive affect subscale of the PANAS, although a trend can be observed at a significance level of p = .097. In the "Nature-based mindfulness training" group, there was a significant difference between T1 and T2 with a mean effect size of d = 0.59, while the difference in the WL group was smaller and not significant. For Negative affect, however, a significant interaction between time and group was found (F = 21.36, df = 1/106.12, p < .001). Here a medium effect was observed in the "Nature-based mindfulness training" group with d = 0.56, while the difference from T1 to T2 in the WL group was not significant. Figure 22 shows the estimates of the scores from the analyses.



Figure 22. Positive and Negative affect in the prevention trial: comparison of the "Nature-based mindfulness training" group and the WL group from T1 (before the first session) to T2 (after the last session)

Looking at the longitudinal analysis of the "Nature-based mindfulness training" group (Fig. 23), no significant effect was found for Positive affect even after three months (T3) compared to the baseline (T1; p = .326). For Negative affect, the effects decreased slightly compared to the second time point immediately after the training (small effect with d = 0.22).

5

4

3

2

1

t1

Means

Negative affect

t2

Nature-based mindfulness training

t3



Figure 23. Affect T1-T2-T3 in the prevention trial

Secondary outcomes

Significant interactions were found for secondary outcomes related to nature-based mindfulness, emotional connectedness to nature, awareness of environmental threat, and internal and external attributions of responsibility for nature conservation. For all these variables, significantly larger effects were observed in the "Nature-based mindfulness training" group than in the WL group. Medium effects also remained at the three-month follow-up.

Non-significant interactions were found for self-compassion, mindfulness, social support, nature conservation attitudes and behaviour, and current contact with nature. However, tendencies towards larger effect sizes for the "Nature-based mindfulness training" group can also be observed for the Mindfulness variable (p = .0998). These trends would need to be further investigated in a study with greater statistical power. It is worth noting that, as in the clinical arms, the effect sizes for current contact with nature were similar. Participants in the "Nature-based mindfulness training" group and the WL group also did not differ in the frequency of contact with nature at T2.

Detailed tables and figures showing the results of the primary and secondary outcomes can be found in Annex C.2.

4.2.2 Process evaluation

The process evaluations show that the individual sessions in the prevention trial were conducted under different situational conditions. Nevertheless, the effects of the intervention can be experimentally demonstrated, indicating the high effectiveness of the offering. The sessions took place from May to November, with most starting in the early morning (9:00 am) and a few in the afternoon (3:00 pm). The average group size was 6 people, with a minimum of 4 and a maximum of 10. Temperatures during the sessions ranged from 1°C to 28°C, and the "nature-based mindfulness training" took place in a variety of weather conditions, including sunshine, clouds, rain, fog, wind and storms.

99 per cent of participants felt comfortable during the sessions (M = 3.78; SD = 0.46), and the same percentage found the intervention effective (M = 3.67; SD = 0.53). Both subjective well-being and perceived effectiveness were very high after the first session and continued to increase over time (see Fig. 24).



Figure 24. Subjective well-being and subjective effectiveness of the intervention over the course of the intervention from the perspective of the participants in the prevention trial.
(1 = I did not feel comfortable; 4 = I felt very comfortable; 1 = I did not find it effective; 4 = I found it very effective.)

The effectiveness, experimentally proven in the questionnaires, is reflected and validated by the subjective experiences of the participants.

4.2.3 Interviews

The results of the interviews from the prevention trial, focusing on the effectiveness of "nature-based mindfulness training", are provided below. The discussion covers the project characteristics necessary for successful implementation from the perspective of the experts, the effectiveness of the intervention in terms of mental, physical and social well-being, nature conservation behaviour, and the assessment of overall effectiveness. This is followed by determinants of effectiveness at the individual and session level, and factors influenced by the implementation site in nature. The category systems with anchor examples, the frequencies of experts who commented on the categories, and the necessary coding rules for these and other secondary constructs can be found in Annex B.2.

Success-relevant project characteristics

Based on the experts' statements, nine characteristics were identified as necessary for the successful implementation of a nature-based intervention in a preventive setting (Fig. 25).



Figure 25. Success-relevant project characteristics of a nature-based intervention in the preventive setting

The most important criterion for the successful implementation of the intervention is seen as the selection of a *suitable location* that is easily accessible (e.g., E9: "Simply finding suitable places in the region is certainly a bit of a challenge.") and at the same time is sufficiently secluded (e.g., E4: "... at the same time it has to be a place that is not frequented by tourists, and then the choice in the southern district becomes very narrow. Because the areas that are easily accessible are often heavily frequented by tourists").

Sufficient *human resources* must also be available (e.g., E4: "... unfortunately, that's not the case with us, we don't have another person who can contribute labour to the project. And it is a disadvantage or simply additional stress if I always have to think of everything for the groups."), and despite a structured organisation of the intervention, it must be possible to *respond flexibly* to situational conditions (e.g., E9: "...they are guided through this intervention very well, so they have many opportunities to ask questions and reflect").

Sufficient *information* must be provided both before and during the intervention (e.g., E4: "I set up email distribution lists for each group and sent out regular information to everyone in advance, and it worked well"), and cooperation with all those involved in the intervention must be maintained (e.g., E4: "It is important to make it very clear to the forest owners, or to request such information in advance, so that we can move to another area if necessary. Or whether the work with wood could perhaps take place at a different time").

The *financial cost* of the intervention has to be considered as an offering for private individuals (e.g., E9: "At the moment it is offered free of charge, or ... the whole thing is offered effectively for free"), as well as the *time commitment* for the intervention (e.g., E11: "But the problem is that they do not have time for the three sessions. This is an obstacle where some dropped out. Where some ... said: A friend also wanted to do it, but three sessions were too much"). Participants must be *recruited* based on inclusion and exclusion criteria (e.g., E4: "... of those who are interested, many do not quite fit into

the sample, especially in terms of age. Because we have this requirement between 18 and 59 and there were a lot of older people involved"). There needs to be a clear *division of labour* within the intervention team.

Effects on mental well-being

According to the experts, the intervention has a variety of potential effects on the mental well-being of the participants (see Fig. 26).





From the outside, the participants appeared *more relaxed and calmer* after the interventions (e.g., E6: "... but also realised that they were already more relaxed emotionally after the implementation"). The experts report that the intervention simultaneously leads to a *directing and focusing of attention* (e.g., E12: "After a few hours in the forest, it is clearly noticeable to everyone, it is clearly palpable ... that a different level of perception has been addressed.") and also contributes to *stress reduction* (e.g., E9: "It's good for them to be able to cope better with all the stress that arises in everyday life because of the coronavirus situation").

Through the intervention, the participants experience *positive emotions* (e.g., E4: "I can say that everyone was in a good mood afterwards.") and *distance themselves from everyday life*. The experts note that the intervention promotes *mindfulness* and counteracts the risk of *burnout*.

Effects on physical well-being

According to the experts, the intervention affects the physical well-being of the participants in two ways (see Fig. 27).



Figure 27. Effects of the intervention on physical well-being in the preventive setting

The intervention leads to *physical activation* of the participants, both during the sessions (E9: "Then there are stretching exercises that you do on the tree."), as well as beyond that (e.g., E12: "... the encouragement and the desire to go out into nature more often ..."). At the same time, the intervention *reduces physical tension and restlessness*.

Effects on social well-being

According to the experts, the intervention also has an impact on the social well-being of the participants (see Fig. 28).



Figure 28. Effects of the intervention on social well-being in the preventive setting

In the course of the intervention, the participants showed very quickly that they were *open* to the experiences (e.g., E9: "... that people were open with each other, very quickly open, that there was such open togetherness right away"). *Group cohesion* is promoted (e.g., E12: "It creates ... a sense of community ..."), and *compassion* for each other is built up (e.g., E4: "I've noticed a change in myself because it's all about compassion"). At the same time, *trust is built* between the participants, mutual *appreciation* is promoted, and there is a general improvement in *social skills*.

Effects on pro-environmental and nature conservation behaviour

The experts' statements indicate that two pro-environmental and nature conservation behaviour are influenced by the intervention (see Fig. 29).



Figure 29. Effects of the intervention on pro-environmental and nature conservation behaviour in the preventive setting

The intervention initiates an intensive *examination of ecological behaviour* in the participants (e.g., E12: "Some ... started to behave differently in nature. This change in attitude was very noticeable to me. ... this encouragement to behave and move differently in nature, or this desire to engage more intensively with nature and become active. So not just 'I'll do it at some point', but 'I'll really go out and do it'."). In addition, a *mindful perception of nature* is encouraged (e.g., E11: "The participants who

were there were mostly very mindful of protecting nature. They were kind of confirmed in their attitude").

Overall effectiveness

As regards the assessment of the overall effectiveness of the intervention, four categories can be found in the experts' statements (see Fig. 30).



Figure 30. Assessment of the overall effectiveness of the intervention in the preventive setting

The *effectiveness* of the intervention is rated as *high* (e.g., E6: "I think that there are few interventions, if I'm talking about psychology, psychotherapy, prevention, that really have such a high effect."), a finding that is consistent with the overall quantitative assessment by the experts, the experimental findings and the subjective assessment of the participants.

In a differentiated view, *individual elements* are perceived as having different effects for different people (e.g., E14: "Although it is true that not every activity is always equally good for everyone."), but this does not diminish the overall effect of the intervention. Their effectiveness is seen both in the *short-term* effects (e.g., E6: "... the effects could be seen right at the end of the day after the intervention") and in the *medium-term* effects (e.g., E14: "... and that people also say that they definitely notice that it has an impact on their everyday lives").

Individual determinants of effectiveness

According to the experts, the participants themselves also influence whether and to what extent the intervention can be effective (see Fig. 31).





In particular, the willingness to *practise* the learned content *independently* (e.g., E4: "... people who are willing to practise daily, for example to do a short meditation every day or to really pause while

eating or walking and pay attention to their own breathing, to their own perception in the present moment, will probably benefit the most.") is considered important.

Openness for the offering (e.g., E4: "... as far as the activities are concerned, ... it is of course good if you are as open as possible to getting involved in activities.") and possible *previous experience* (e.g., E12: "Someone who already has more experience has a wider range of perception.") also influence the effectiveness of the intervention. The *physical condition* (E13: "... someone who is very overweight or has severely restricted mobility would not even register for the course.") and the degree of *introversion and extraversion* must also be taken into account.

Intervention-related determinants of effectiveness

According to the experts' statements, in addition to the individual determinants of effectiveness, factors on the intervention level should also be considered (see Fig. 32).



Figure 32. Determinants of intervention effectiveness on the session level in the preventive setting

This starts with possible *disturbances during the offering* (e.g., E4: "... and that was very loud and that is something that really disturbed the group. Even the participants said that it disturbed them a lot."), which might reduce effectiveness. *Attending the intervention at least twice* is considered necessary (e.g., E9: "The second time it was generally quieter").

On the group dynamic level, the *atmosphere in the group* (e.g., E9: "... that this group is also so uncomplicated and that it was so open to each other right away.") and the *cohesiveness of the group* (e.g., E9: "It is of course a closed group, which is ... very good since you are always together with the same people for these four sessions.") are experienced as important factors contributing to effectiveness.

Effectiveness of the natural environment

Finally, according to the experts, the setting of the intervention, i.e., the natural environment, influences the extent to which the intervention is effective (see Fig. 33).



Figure 33. Determinants of the intervention effectiveness of the natural environment in the preventive setting

High importance is attached to nature as the site of intervention (e.g., E9: "I simply feel different when I am ... in a forest or on a meadow, compared to when I am in a room"). This is because a comprehensive *sensory experience* is made possible there (e.g., E4: "Just feeling rain, cold, heat, sun on your skin, consciously smelling, hearing, even touching, it would be very time-consuming if you wanted to do that indoors.") and *memories* of the past are activated (e.g., E14: "In addition, past experiences that came up during the activities are also discussed. For example, childhood memories, specific experiences in nature").

The *diversity* of nature also contributes to effectiveness (e.g., E12: "A place that shows different trees, that has both open spaces for activities and the opportunity to be very close to trees"). In addition, the *activities' connection to nature* (e.g., E6: "...if only because many activities cannot be carried out without nature.") and the *emotional connection* with nature (E2: "...this aspect that we are part of nature.") promote effectiveness.

The intervention is perceived as being feasible and effective *regardless of* site (e.g., E4: "So of course I need a space, an area, where the group can be, a meadow, a forest, a bit of variety in the landscape. But I don't think it would make any difference whether this training took place near Freilassing or in the Rhön."), as long as nature is experienced as a *place of retreat* (e.g., E12: "A place ... that creates a sense of security."), is *naturally* available in its *design*, is *aesthetically pleasing* and offers a *low-stimulus environment*.

4.3 Subgroup analyses

Subgroup analyses were also calculated for the preventive arms. The report here focuses on the primary outcomes, the two subscales of Positive and Negative affect of the PANAS. In addition, the significance level for the subgroup analyses of the preventive samples was set at p = .01 (Bonferroni correction for five subgroup analyses: 0.05/5 = 0.01).

Childhood residence

The Childhood residence variable was used as a binary variable (urban/rural) in the analyses. A significant interaction of Time*Group*Childhood residence was observed for Positive affect (F = 11.226, df = 1/115.028, p = .001). Post-hoc comparisons revealed a significant increase in Positive affect only in the "Nature-based mindfulness training" group among the group of people who had spent their childhood in a rural environment. However, this effect diminished significantly in the longitudinal analysis. For Negative affect, the effect was not significant when the significance level was adjusted (F = 4.45, df = 1/120.29, p = .037). However, the direction of the effect was similar to that for Positive affect: individuals from rural backgrounds experienced a greater decrease in Negative affect. In the longitudinal analysis, the significances exceeded the critical level of p = .01, so no significant effects can be claimed here.

Gender

In the prevention group, the Gender variable was also divided into three categories, with no individuals identifying as "diverse". There was no significant interaction with gender for the Positive and Negative affect from T1 to T2. In the longitudinal analysis, a significant interaction was observed only for Positive affect (F = 4.36, df = 3/113.17, p = .006). The analysis of the estimates for the individual measurements showed a decrease in Positive affect at the follow-up time point T3 back to baseline only for female participants.

Previous experience

Participants with previous experience did not differ significantly from participants without previous experience in the primary outcomes of Positive and Negative affect. This was observed both in the group comparison of "Nature-based mindfulness training" vs. WL and in the longitudinal analysis of the "Nature-based mindfulness training" group.

Age

Younger and older participants did not differ significantly in the effects of the intervention on Positive and Negative affect. This was true for both the group comparison of "Nature-based mindfulness training" vs. WL and in the longitudinal analysis of the "Nature-based mindfulness training" group.

Children

Participants with children did not differ from participants without children in the effectiveness of the "Nature-based mindfulness training", both in comparison to the WL group and in the longitudinal analysis of the "Nature-based mindfulness training" group.

5 Discussion of the results

5.1 Discussion of the results in relation to the research questions

The Green Care study, a mindfulness- and relaxation-based nature intervention for depressed inpatients in psychosomatic rehabilitation centres and healthy adults, aimed to compare the effects of a nature intervention (four sessions of a "Nature-based relaxation offering" in the UNESCO Biosphere Reserve Rhön or three sessions of a "Nature-based mindfulness training" in the UNESCO Biosphere Reserve Berchtesgadener Land) with standard treatment plus a Waiting List control group (TAU+WL) in the clinical setting and with a Waiting List control group (WL) in the preventive setting (three sessions of a "Nature-based mindfulness training" in the UNESCO Biosphere Reserve Berchtesgadener Land).

Clinical setting

In the clinical setting, patients were divided into two groups: a Green Care group, which received the interventions in addition to their standard treatment, and a control group, which did not receive the intervention during their inpatient stay, but were offered a short one- or two-day intervention after the second data collection before discharge. Both participating centres were located in UNESCO biosphere areas, ensuring that all patients had access to nature in the form of parks and forests. This ensured that the effects were not solely due to different levels of contact with nature.

Comparison of the groups at admission (T1) and at discharge (T2) shows that the Green Care intervention had a greater positive effect on patients' mood compared with standard treatment, with large effects in comparison to medium to large effects in the TAU+WL group. These effects were found in the primary outcomes of the study, the scales of Positive and Negative affect of the PANAS, and were independent of the study centre. Sensitivity analyses showed that these effects did not depend on the level of depression, gender, age, previous experience with nature interventions or mindfulness training, childhood residence, or whether or not the patients had children.

In terms of secondary outcomes, self-compassion showed a significant effect in the health-related variables, similar to the PANAS. Depression and mindfulness did not change significantly, but the direction of the effects was similar to the primary outcomes. With mindfulness, it was possible to identify a trend in the sense that the significance level was below p = .010. Importantly, contact with nature did not change significantly between groups, and there was no notable trend between groups. This suggests that contact with nature alone, without professional guidance, may not be a critical factor in the effectiveness of the intervention.

In terms of nature-related variables, a significant effect was observed in patients' connectedness to nature. Differences in nature-related mindfulness, nature conservation attitudes and behaviour were not statistically significant. This may be related to the fact that patients in an inpatient setting have limited opportunities outside of daily life to demonstrate nature conservation behaviour or to establish behavioural changes. However, the results show that connectedness to nature can be significantly improved by the intervention. Previous research suggests that connectedness to nature is a good predictor of future nature conservation attitudes and behaviour (cf. Kals & Nisbet, 2019).

Patients in the Green Care group were contacted for a follow-up assessment three months after the intervention. The low drop-out rate of 11 per cent – with 92 out of 103 people responding – demonstrates the high level of patient interest in the Green Care programme. The data show that the effects were reduced compared to discharge (T2), but remained high compared to admission (T1). Sensitivity analyses showed that patients with higher levels of depression did not show significant differences in affect from T2 to T3, suggesting that this group of patients continued to benefit from the Green Care intervention after three months. Patients with high levels of distress, as indicated by high scores on the depression scale, are often challenging in therapy and treatment. The study provides promising results that this group of patients can benefit from the programme in the long term.

Another variable that moderated the longitudinal analysis results was childhood residence (urban vs. rural). Patients who had spent their childhood in an urban environment did not show a significant increase in Negative affect three months after discharge compared with immediately after discharge.

Preventive setting

In the preventive setting, two groups were also established: the Green Care group, which received three sessions of nature-based mindfulness training, and the Waiting List control group (WL), which received no intervention initially but could participate in a nature-based intervention session after completing the questionnaires at T2. This part of the study was conducted exclusively in the UNESCO Biosphere Reserve Berchtesgadener Land. Therefore, the sample size in the preventive setting was smaller, resulting in less statistical power in the analyses. This means that existing effects are recognised as significant with lower likelihood.

The comparison between the Green Care group and the WL group at T1 and T2 shows that mood improved in the Green Care group. There was a significant difference in Negative affect compared to the WL group. The comparison of Positive affect did not yield significant results in the analysis, but there is a tendency that participants in the Green Care group benefited more than those in the WL group. The longitudinal analysis confirms these trends. Subgroup analyses show that there is indeed a significant difference in Positive affect, but only among people who grew up in an urban environment. Men also appear to benefit more from the intervention in the longitudinal analysis, as they continue to experience more Positive affect. The data suggest that the Green Care programme seems to be particularly suitable for people from urban areas in a preventive setting. The effectiveness is somewhat lower for people from rural areas, but still noticeable.

For secondary outcomes, significant effects were found, especially for nature-related variables, but not for self-compassion. For mindfulness variable, there are trends that may not have reached the point of significance due to the slightly smaller sample size. The consistent direction of the (nonsignificant) effects showing a greater effectiveness of the Green Care intervention relative to the WL group supports this assumption. In the case of nature-related variables, significant effects were found for nature-related mindfulness, emotional connectedness to nature, awareness of environmental threats, and internal and external attributions of responsibility, but not in the case of nature conservation attitudes and behaviour. However, no significant effect on nature conservation mindsets can be reported here as well. Environmental psychology has been able to demonstrate that emotional

attachment to nature is a central factor for engagement and behaviour in the context of nature conservation, while internal attribution of responsibility, which means accepting personal responsibility for nature conservation, can be viewed as another central predictor (cf. chapter 2).

Process evaluation

Comprehensive process evaluations are available for the sessions in both the clinical and preventive settings, including data on session conditions (environment, weather, temperature, group size, incidents) as well as participant feedback on subjective well-being and perceived effectiveness. The analysis shows that participants in all groups felt very comfortable and rated the intervention as highly effective. These ratings were high from the first session and increased as the interventions progressed. The longer a person took part in Green Care, the more they appreciated its effectiveness and their well-being. There is considerable variability in weather and temperature conditions, but the analyses show that there is no measurable relationship between these conditions and subjective ratings of effectiveness and well-being. This means that Green Care offerings are effective in a wide range of weather conditions. No differences were found in terms of the natural environment in which the interventions took place. However, it should be noted that in almost all sessions a forest was visited as a natural environment (mixed forest, beech forest, riparian forest, pine forest), so there is not enough data for a more in-depth analysis. This would require further research.

Qualitative results

The interviews in both the clinical and preventive settings provide additional, differentiated insights into the success factors of the programme. The guidelines covered a wide range of relevant topics, starting with structural characteristics, themes of effects on mental, physical and social well-being, on nature conservation aspects, as well as on the global level. Factors influencing effectiveness at the individual level, the intervention itself and the natural environment were also addressed. In terms of the global level, the interviews with the experts show a similar direction to the quantitative data, indicating high effectiveness and acceptance of the interventions. The subjective perception of the experts is largely consistent with the experimentally demonstrated results. In addition, the interviewees identified highly significant factors that are necessary for successful implementation of the interventions in both clinical and preventive settings. On the structural level, these include successful collaboration with treating professionals in the clinical setting, equipment and personnel resources, and the availability of suitable natural spaces for the intervention. On the individual level, willingness to practice independently and openness to nature were frequently mentioned. With regard to the intervention itself, it was seen that not only the group, but also above all the scope of the intervention turned out to be a significant factor. According to the experts, a more significant reduction of the offering – in the project three to four sessions were offered – is not recommended.

Regarding possible discrepancies between qualitative and quantitative findings, it should be noted that qualitative designs allow for inductive exploration and comprehensive analysis of research field, delving deep into the subject matter. Quantitative studies, by contrast, use standardised measurements to test theories and hypotheses deductively, with the aim, among other things, of explaining causal relationships. Combining these two methods, as was done in the present evaluation

research, allows the strengths of both research approaches to be combined and the weaknesses of each to be compensated for by the other. Findings from qualitative and quantitative studies that allow for different conclusions, such as the effectiveness of the intervention on nature-related mindfulness, are therefore expected in multimethod study designs and can be seen as a reason for further research.

5.2 Discussion of strengths and limitations

Strengths

The study has several strengths that are worth highlighting. While there has been some research on the effects of nature-based offerings investigated under controlled conditions, the present study took place in the real-world context of psychosomatic clinics and adult education centres. This allows a better assessment of the feasibility and effectiveness of the offering in terms of ecological validity.¹ Overall, the study demonstrated the feasibility of nature-based interventions in both clinical and preventive settings. In the clinical setting, the programmes were successfully integrated into the treatment plan. In the preventive setting, the Green Care offering received a strong positive response and attracted many participants. The study found a high level of interest in the programmes, as evidenced by the very low dropout rate in the treatment or training groups and the high participation rate in the Waiting List control groups.

The offerings, both clinical and preventive, took place in easily accessible and well-located natural settings, mostly in forest areas. This suggests that the interventions could be applied in other settings. The specific design of the interventions allows for flexibility if professional planning and guidance is provided.

The results of the summative and process evaluations, as well as the results of the interviews by experts, provide a clear picture of the effectiveness of the interventions through a multi-method approach. Participants in all groups expressed a high level of satisfaction. Interviews and experience of implementing the programmes in the clinical setting also showed a high level of interest and acceptance among clinical staff for the programme. Subgroup analyses also accounted for potential influential factors (weather, previous experience, children, etc.), which did not affect effectiveness.

A 3-month follow-up of the Green Care groups showed slight decreases in some variables, but still indicated the stability of the results, especially for highly distressed patients in the clinical setting.

Limitations

Despite all efforts to create a practical and attractive offering for participants, the study has some limitations.

Firstly, it was not a randomised trial, but a controlled trial. In the clinical arm, because all patients interacted and shared information during treatment in the clinic, individual randomisation was not possible. To control for allocation to the study arms, a propensity score was used in all analyses to

¹ Ecological validity is understood as the validity of psychological statements for the real-life context and everyday event (Bortz & Döring, 2016).

statistically adjust for differences between the groups. In addition, the prospectively registered study clearly distinguished between primary and secondary outcomes, ensuring transparency in the criteria for study success, which could not be influenced retrospectively by arbitrary selection of significant *p*-values.

Secondly, the two treatment programmes differed slightly at the sites, with a greater focus on relaxation and mindfulness and slightly different techniques used in each session. In addition, the clinical arms in both centres had to take into account the circumstances of the clinics (closed vs. open groups, four vs. three treatment sessions, weather-related restrictions). However, the analyses showed no significant differences between the centres, confirming the effectiveness of both programmes.

Thirdly, no follow-up data could be collected for the control groups. These groups were implemented after the Green Care interventions, so a follow-up study would not have been possible until after the end of the project. In addition, the Waiting List offering could only have been implemented later, after the follow-up measurement, which would not have been realistic given the long distances that many patients and participants had to travel.

Finally, for organisational, environmental and ethical reasons, it was not possible to collect additional data on clinical patients and preventive participants, such as detailed clinical differential diagnoses, health status, and experiences and personal practices outside the offering and after the training. As contact with nature decreased slightly in the Green Care groups at the time of the follow-up, it can be assumed that many participants only minimally integrated the activities into their daily lives. The application of the practical activities in daily life should receive special attention in further training, especially in the clinical context.

6 Conclusion

The study, with its comprehensive parts, showed that the interventions were very practicable and feasible in real-world settings, despite numerous challenges such as pandemic-related measures and associated restrictions:

- Good feasibility in clinical practice was confirmed. On the one hand, integration into treatment processes seemed challenging at first, as clinical routines left little room for it. On the other hand, the high level of commitment of those responsible in the clinics demonstrated acceptance and feasibility.
- The proximity of the natural environments visited in the offerings is another important point. In order to implement a successful Green Care programme in the long term, it is not absolutely necessary to undertake extensive trips that require additional planning and supervision.
- The study has shown that certain prerequisites are important for the successful implementation of the programme. Contact with nature alone does not have an effect;

rather, a professionally managed Green Care programme is essential. In the trials, participants in the Wait List control groups and the Green Care groups had similar levels of contact with nature. If contact with nature alone were effective, no group differences would have been expected. It is therefore likely that high-quality and professionally led programmes are very important for the effectiveness of the intervention.

- The individualised design of the training allows for flexibility; and the programmes were shown to be effective at both sites. However, it is always important to align the structure and focus of the training with theoretical knowledge and empirical scientific evidence.
- The experiences from the offerings, especially those of the actors at the two sites, can and should inform the development of multiplier programmes. This could be achieved, for example, by standardising the programme and training the trainers.

The "Green Care – Nature and Mental Health" study has shown that the offerings implemented are innovative and effective. From a scientific point of view, the continuation of the "Green Care | Nature and Mental Health" project is strongly recommended.

Works cited

- Abraham, A., Sommerhalder, K., Bollinger-Salzmann, H., & Abel, T. (2007). *Landschaft und Gesundheit: Das Potential einer Verbindung zweier Konzepte.* Institut für Sozial- und Präventivmedizin. https://doi.org/10.7892/boris.73684
- Annerstedt, M., & Währborg, P. (2011). Nature-assisted therapy: Systematic review of controlled and observational studies. *Scandinavian Journal of Public Health*, 39(4), 371-388. https://doi.org/10.1177/1403494810396400
- Astles, B. L. (2015). *The Impact of Outdoor Environments on Health and Well-being of Residents in Long-Term Care Facilities: A Review of the Literature* [master's thesis]. Simon Fraser University Waterloo.
- Bailey, A. W., & Kang, H.-K. (2022). Walking and Sitting Outdoors: Which Is Better for Cognitive Performance and Mental States?. *International Journal of Environmental Research and Public Health*, 19(24), 1-11. https://doi.org/10.3390/ijerph192416638
- Ballew, M. T., & Omoto, A. M. (2018). Absorption: How nature experiences promote awe and other positive emotions. *Ecopsychology*, *10*(1), 26-35. https://doi.org/10.1089/eco.2017.0044
- Berman, M. G., Jonides, J., & Kaplan, S. (2008). The cognitive benefits of interacting with nature. *Psychological Science*, *19*(12), 1207–1212. https://doi.org/10.1111/j.1467-9280.2008.02225.x
- Berto, R. (2005). Exposure to restorative environments helps restore attentional capacity. *Journal of Environmental Psychology*, *25*(3), 249-259. https://doi.org/10.1016/j.jenvp.2005.07.001
- Beyer, K. M., Kaltenbach, A., Szabo, A., Bogar, S., Nieto, F. J., & Malecki, K. M. (2014). Exposure to neighborhood green space and mental health: Evidence from the survey of the health of Wisconsin. *International Journal of Environmental Research and Public Health*, *11*(3), 3453–3472. https://doi.org/10.3390/ijerph110303453
- Bielinis, E., Jaroszewska, A., Łukowski, A., & Takayama, N. (2019). The Effects of a Forest Therapy Programme on Mental Hospital Patients with Affective and Psychotic Disorders. *International Journal of Environmental Research and Public Health*, 17(1), 1-10. https://doi.org/10.3390/ijerph17010118
- Blum, B. (2019). Naturschützende Engagementbereitschaften und Verhaltensweisen: eine psychologische Studie zu verantwortungsbezogenen Motivatoren und Barrieren [master's thesis]. Katholische Universität Eichstätt-Ingolstadt.
- Bogner, A., & Menz, W. (2002). Expertenwissen und Forschungspraxis: Die modernisierungstheoretische und die methodische Debatte um die Experten. In A. Bogner, B. Littig, & W. Menz (Eds.), *Das Experteninterview: Theorie, Methode, Anwendung* (pp. 7–29). VS Verlag für Sozialwissenschaften. https://doi.org/10.1007/978-3-322-93270-9_1

- Bowler, D. E., Buyung-Ali, L. M., Knight, T., & Pullin, A. (2010). A systematic review of evidence for the added benefits to health of exposure to natural environments. *BMC Public Health*, *10*(456), 1-10. https://doi.org/10.1186/1471-2458-10-456
- Bratman, G. N., Daily, G. C., Levy, B. J., & Gross, J. J. (2015). The benefits of nature experience: Improved affect and cognition. *Landscape and Urban Planning*, *138*, 41-50. https://doi.org/10.1016/j.landurbplan.2015.02.005
- Britton, E., Kindermann, G., Domegan, C., & Carlin, C. (2020). Blue care: a systematic review of blue space interventions for health and wellbeing. *Health promotion international*, 35(1), 50–69. https://doi.org/10.1093/heapro/day103
- Bröderbauer, D. (2015). Naturerleben und Gesundheit: Eine Studie zur Auswirkung von Natur auf das menschliche Wohlbefinden unter besonderer Berücksichtigung von Waldlebensräumen.
 Naturfreunde Internationale.
 https://www.bundesforste.at/uploads/publikationen/WasserWege_Gesundheit.pdf
- Brown, K. W., & Ryan, R. M. (2009). The mindfulness attention awareness scale (MAAS). *Acceptance and commitment therapy. Measures Package*, *82*.
- Bruckbauer, S., Strubel, I. T., & Kals, E. (2022). *From ecological to sustainable identity: validation of the Sustainable Identity Scale and the impact of identity on sustainable behavior* [Manuscript submitted for publication].
- Bundespsychotherapeutenkammer. (2018). *Ein Jahr nach der Reform der Psychotherapie-Richtlinie Wartezeiten 2018*. BPtK. https://www.bptk.de/publikationen/bptk-studie/
- Burcusa, S. L., & Iacono, W. G. (2007). Risk for recurrence in depression. *Clinical Psychology Review,* 27(8), 959–985. https://doi.org/10.1016/j.cpr.2007.02.005
- Cackowski, J. M., & Nasar, J. L. (2003). The restorative effects of roadside vegetation implications for automobile driver anger and frustration. *Environmental Behavior, 35,* 736-751.

Cervinka, R., Höltge, J., Pirgie, L., Schwab, M., Sudkamp, J., Haluza, D., Arnberger, A., Eder, R., & Ebenberger, M. (2014). Zur Gesundheitswirkung von Waldlandschaften: Green Public Health - Benefits of Woodlands on Human Health and Wellbeing [BFW- Bericht 147].
Bundesforschungszentrum für Wald.
http://bfw.ac.at/cms_stamm/GreenCareWald/pdf/BFW_Bericht147_2014_Green
PublicHealth.pdf

- Choi, H., Hahm, S.-C., Jeon, Y.-H., Han, J.-W., Kim, S.-Y., & Woo, J.-M. (2021). The Effects of Mindfulness-Based Mandala Coloring, Made in Nature, on Chronic Widespread Musculoskeletal Pain: Randomized Trial. *Healthcare*, 9(6), 1-12. https://doi.org/10.3390/healthcare9060642
- Chun, M. H., Chang, M. C., & Lee, S.-J. (2017). The effects of forest therapy on depression and anxiety in patients with chronic stroke. *The International Journal of Neuroscience*, *127*(3), 199–203. https://doi.org/10.3109/00207454.2016.1170015

- Coley, R. L., Sullivan, W. C., & Kuo, F. E. (1997). Where Does Community Grow?: The Social Context Created by Nature in Urban Public Housing. *Environment and Behavior, 29*(4), 468–494. https://doi.org/10.1177/001391659702900402
- Corazon, S. S., Sidenius, U., Poulsen, D. V., Gramkow, M. C., & Stigsdotter, U. K. (2019). Psycho-Physiological Stress Recovery in Outdoor Nature-Based Interventions: A Systematic Review of the Past Eight Years of Research. *International Journal of Environmental Research and Public Health, 16*(10), 1-21. https://doi.org/10.3390/ijerph16101711
- Corazon, S. S., Sidenius, U., Vammen, K. S., Klinker, S. E., Stigsdotter, U. K., & Poulsen, D. V. (2018).
 The Tree Is My Anchor: A Pilot Study on the Treatment of BED through Nature-Based Therapy.
 International Journal of Environmental Research and Public Health, 15(11).
 https://doi.org/10.3390/ijerph15112486
- Cordes, A., Herrmann-Lingen, C., Büchner, B., & Hessel, A. (2009). Repräsentative Normierung des ENRICHD-Social-Support-Instrument (ESSI)-Deutsche Version. *Klinische Diagnostik und Evaluation*, *2*(1), 16-32.
- Coventry, P. A., Brown, J. E., Pervin, J., Brabyn, S., Pateman, R., Breedvelt, J., Gilbody, S., Stancliffe, R., McEachan, R., & White, P. L. (2021). Nature-based outdoor activities for mental and physical health: Systematic review and meta-analysis. SSM - Population Health, 16, 100934. https://doi.org/10.1016/j.ssmph.2021.100934
- Cox, D. T., Shanahan, D. F., Hudson, H. L., Plummer, K. E., Siriwardena, G. M., Fuller, R. A., Anderson,
 K., Hancock, S., & Gaston, K. J. (2017). Doses of neighborhood nature: The benefits for mental health of living with nature. *BioScience*, 67(2), 147-155. https://doi.org/10.1093/biosci/biw173
- Daniels, S., Clemente, D. B. P., Desart, S., Saenen, N., Sleurs, H., Nawrot, T. S., Malina, R., & Plusquin, M. (2022). Introducing nature at the work floor: A nature-based intervention to reduce stress and improve cognitive performance. *International journal of hygiene and environmental health*, 240, 113884. https://doi.org/10.1016/j.ijheh.2021.113884
- DeGEval Gesellschaft für Evaluation e.V. (2016). *Standards für Evaluation* (Erste Revision, Kurzfassung). Verfügbar unter: https://www.degeval.org/ fileadmin/DeGEval-Standards/2019_07_10_DeGEval_Standards__Kurzfassung.pdf
- Deutsche Psychotherapeuten Vereinigung e.V. (2021). *Report Psychotherapie 2021*. Wissenschaft und Forschung.
- Dienemann, K. (2020). Wirkungen der Natur auf die Psyche. In C. Gans, K. Dienemann, A. Hume, & A. Lorino (Eds.), *Arbeitsraum Natur: Handbuch für Coaches, Therapeuten, Trainer und Organisationen* (pp. 31-37). Springer.
- Djernis, D., Lerstrup, I., Poulsen, D., Stigsdotter, U., Dahlgaard, J., & O'Toole, M. (2019). A systematic review and meta-analysis of nature-based mindfulness: Effects of moving mindfulness training into an outdoor natural setting. *International Journal of Environmental Research and Public Health, 16*(17). https://doi.org/10.3390/ijerph16173202

- Dresing, T., & Pehl, T. (Eds.). (2015). *Praxisbuch Interview, Transkription & Analyse. Anleitungen und Regelsysteme für qualitativ Forschende* (6th edition). Dr. Dresing und Pehl GmbH.
- Engel, G. L. (1977). The need for a new medical model: A challenge for biomedicine. *Science*, *196*(4286), 129–136. https://doi.org/10.1126/science.847460
- Ensinger, K. (2016). Achtsamkeit, Naturerleben und die Erfahrung von Erholung. *Umweltpsychologie*, 20(2), 95–111.
- Europäische Kommission (2005). Die psychische Gesundheit der Bevölkerung verbessern –
 Entwicklung einer Strategie für die Förderung der psychischen Gesundheit in der Europäischen Union. *KOM(200), 484.*https://ec.europa.eu/health/archive/ph_determinants/life_style/mental/green_paper/me ntal_gp_de.pdf
- Flade, A. (2018). Zurück zur Natur? Erkenntnisse und Konzepte der Naturpsychologie. Springer Fachmedien.
- Flick, U. (2017). *Qualitative Sozialforschung. Eine Einführung* (8th edition). Rowohlt.
- Gittins, H., Dandy, N., Wynne-Jones, S., & Morrison, V. (2023). "It's opened my eyes to what's out there": How do nature-based interventions influence access to and perceptions of the natural environment?. *Wellbeing, Space and Society, 4*, 100125. https://doi.org/10.1016/j.wss.2022.100125
- Grassini, S. (2022). A Systematic Review and Meta-Analysis of Nature Walk as an Intervention for Anxiety and Depression. *Journal of Clinical Medicine*, *11*(6). https://doi.org/10.3390/jcm11061731
- Han, J.-W., Choi, H., Jeon, Y.-H., Yoon, C.-H., Woo, J.-M., & Kim, W. (2016). The Effects of Forest Therapy on Coping with Chronic Widespread Pain: Physiological and Psychological Differences between Participants in a Forest Therapy Program and a Control Group. *International Journal of Environmental Research and Public Health, 13*(3). https://doi.org/10.3390/ijerph13030255
- Hartig, T., Evans, G. W., Jammer, L. D., Davis, D. S., & Gärling, T. (2003). Tracking restoration in natural and urban field settings. *Journal of Environmental Psychology*, 23(2), 109-123. https://doi.org/10.1016/S0272-4944(02)00109-3
- Hartig, T., Mitchell, R., De Vries, S., & Frumkin, H. (2014). Nature and Health. *Annual Review of Public Health*, *35*(1), 207-228. https://doi.org/10.1146/annurev-publhealth-032013-182443
- Haubenhofer, D. K., Elings, M., Hassink, J., & Hine, R. E. (2010). The development of green care in western European countries. *Explore*, 6(2), 106–111. https://doi.org/10.1016/j.explore.2009.12.002
- Heise, P., & Hallermayr, S. (2022). Wirkungen von Natur auf den Menschen. In P. Heise, & S.
 Hallermayr (Eds.), *Grüne Stadt Gesunder Mensch* (pp. 11–37). Springer.
 https://doi.org/10.1007/978-3-662-65317-3_2

- Ho, P. L., Li, T. W., Liu, H., Yeung, T. F., & Hou, W. K. (2022). Testing a New Protocol of Nature-Based Intervention to Enhance Well-Being: A Randomized Control Trial. *International Journal of environmental research and public health*, 19(7). https://doi.org/10.3390/ijerph19073931
- Huber, D., Freidl, J., Pichler, C., Bischof, M., Kiem, M., Weisböck-Erdheim, R., Squarra, G., de Nigris,
 V., Resnyak, S., Neberich, M., Bordin, S., Zechner, R., & Hartl, A. (2023). Long-Term Effects of
 Mountain Hiking vs. Forest Therapy on Physical and Mental Health of Couples: A Randomized
 Controlled Trial. *International Journal of Environmental Research and Public Health, 20*(2).
 https://doi.org/10.3390/ijerph20021469
- Hyvönen, K., Salonen, K., Paakkolanvaara, J.-V., Väkeväinen, P., & Korpela, K. (2023). Effects of nature-based intervention in the treatment of depression: A multi-center, randomized controlled trial. *Journal of Environmental Psychology*, 85, 101950. https://doi.org/10.1016/j.jenvp.2022.101950
- Irvine, K. N., Marselle, M. R., Melrose, A., & Warber, S. L. (2020). Group Outdoor Health Walks Using Activity Trackers: Measurement and Implementation Insight from a Mixed Methods Feasibility Study. International Journal of Environmental Research and Public Health, 17(7). https://doi.org/10.3390/ijerph17072515
- Joschko, L., Pálsdóttir, A. M., Grahn, P., & Hinse, M. (2023). Nature-Based Therapy in Individuals with Mental Health Disorders, with a Focus on Mental Well-Being and Connectedness to Nature-A Pilot Study. *International Journal of Environmental Research and Public Health, 20*(3). https://doi.org/10.3390/ijerph20032167
- Kals, E. (Eds). (1998). Umwelt und Gesundheit: Verknüpfung ökologischer und gesundheitlicher Ansätze. Psychologie Verlags Union.
- Kals, E. (2014). Affective Connection to Nature. In A. C. Michalos (ed.), *Encyclopedia of quality of life and well-being research* (p.83-88). Springer.
- Kals, E., & Nisbet, E. (2019). Affective connection to nature. In A. C. Michalos (Ed.), *Encyclopedia of quality of life and well-being research* (2nd edition, pp. 83-88). Springer.
- Kals, E., Schumacher, D., & Montada, L. (1999). Emotional Affinity toward Nature as a Motivational Basis to Protect Nature. *Environment and Behavior*, *31*(2), 178–202. https://doi.org/10.1177/00139169921972056
- Kals, E., Strubel, I. T., & Hellbrück, J. (2023). Umweltpsychologie. Springer.
- Kaplan, R., & Kaplan, S. (1989). *The Experience of Nature: a psychological perspective.* Cambridge University Press.
- Keenan, R., Lumber, R., Richardson, M., & Sheffield, D. (2021). Three good things in nature: a naturebased positive psychological intervention to improve mood and well-being for depression and anxiety. *Journal of Public Mental Health, 20*(4), 243–250. https://doi.org/10.1108/JPMH-02-2021-0029

- Kellert, S. R., & Wilson, E. O. (1993). *The biophilia hypothesis*. Island Press.
- Keniger, L. E., Gaston, K. J., Irvine, K. N., & Fuller, R. A. (2013). What are the Benefits of Interacting with Nature?. *International Journal of Environmental Research and Public Health*, 10(3), 913-935. https://doi.org/10.3390/ijerph10030913
- Kingsley, J., & Townsend, M. (2006). 'Dig in' to social capital: Community gardens as mechanisms for growing urban social connectedness. Urban Policy and Research, 24(4), 525-537. https://doi.org/10.1080/08111140601035200
- Kotera, Y., & Fido, D. (2022). Effects of Shinrin-Yoku Retreat on Mental Health: a Pilot Study in Fukushima, Japan. *International Journal of Mental Health and Addiction, 20*(5), 2652–2664. https://doi.org/10.1007/s11469-021-00538-7
- Kotera, Y., Richardson, M., & Sheffield, D. (2022). Effects of Shinrin-Yoku (Forest Bathing) and Nature Therapy on Mental Health: a Systematic Review and Meta-analysis. *International Journal of Mental Health and Addiction, 20*(1), 337–361. https://doi.org/10.1007/s11469-020-00363-4
- Kroenke, K., Spitzer, R. L., & Williams, J. B. (2001). The PHQ-9: validity of a brief depression severity measure. *Journal of general internal medicine*, *16*(9), 606-613. https://doi.org/10.1046/j.1525-1497.2001.016009606.x.
- Kuo, M. (2015). How might contact with nature promote human health? Promising mechanisms and a possible central pathway. *Frontiers in Psychology*, *6*, 1093. https://doi.org/10.3389/fpsyg.2015.01093
- Kuo, F. E., & Sullivan, W. C. (2001). Aggression and violence in the inner city: Effects of environment via mental fatigue. *Environment and Behavior*, 33, 543–571. https://doi.org/10.1177/00139160121973124
- Kweon, B.-S., Sullivan, W. C., & Wiley, A. R. (1998). Green Common Spaces and the Social Integration of Inner-City Older Adults. *Environment and Behavior*, *30*(6), 832–858. https://doi.org/10.1177/001391659803000605
- Leucht, S., Hierl, S., Kissling, W., Dold, M., & Davis, J. M. (2012). Putting the efficacy of psychiatric and general medicine medication into perspective: review of meta-analyses. *The British Journal of Psychiatry, 200*(2), 97–106. https://doi.org/10.1192/bjp.bp.111.096594
- Ma, J., Williams, J., Morris, P. G., & Chan, P. S. W. Y. (2022). Effectiveness of mindful walking intervention in nature on sleep quality and mood among university student during Covid-19: A randomised control study. *Explore* [in press]. https://doi.org/10.1016/j.explore.2022.08.004
- Maas, J., Verheij, R. A., de Vries, S., Spreeuwenberg, P., Schellevis, F. G., & Groenewegen, P. P. (2009). Morbidity is related to a green living environment. *Journal of epidemiology and community health*, 63(12), 967–973. https://doi.org/10.1136/jech.2008.079038
- Mackinnon, A., Jorm, A. F., Christensen, H., Korten, A. E., Jacomb, P. A., & Rodgers, B. (1999). A short form of the Positive and Negative Affect Schedule: Evaluation of factorial validity and invariance

across demographic variables in a community sample. *Personality and Individual differences*, 27(3), 405-416.

- Marschall, J., Hildebrandt, S., Kleinlercher, K.-M., & Nolting, H. D. (2020). *DAK- Gesundheitsreport* 2020: Beiträge zur Gesundheitsökonomie und Versorgungsforschung (Band 33). DAK Gesundheit.
- Martin, L., White, M. P., Hunt, A., Richardson, M., Pahl, S., & Burt, J. (2020). Nature contact, nature connectedness and associations with health, wellbeing and pro-environmental behaviours. *Journal of Environmental Psychology, 68*, 101389. https://doi.org/10.1016/j.jenvp.2020.101389
- Mayring, P. (2016). *Einführung in die qualitative Sozialforschung. Eine Anleitung zu qualitativem Denken* (revised edition). Beltz.
- McMahan, E. A., & Estes, D. (2015). The effect of contact with natural environments on positive and negative affect: A meta-analysis. *The Journal of Positive Psychology*, *10*(6), 507-519. https://doi.org/10.1080/17439760.2014.994224
- Morita, E., Imai, M., Okawa, M., Miyaura, T., & Miyazaki, S. (2011). A before and after comparison of the effects of forest walking on the sleep of a community-based sample of people with sleep complaints. *BioPsychoSocial Med*, *5*(13).
 https://doi.org/10.1186/1751-0759-5-13
- Müller, M. M., Hemmer, I., & Trappe, M. (Eds.). (2014). *Nachhaltigkeit neu denken. Rio + X: Impulse für Bildung und Wissenschaft*. Oekom Verlag.
- Müller, M., Kals, E., & Pansa, R. (2009). Adolescents' emotional affinity towards nature: A crosssocietal study. Special issue: children and nature. *Journal of Developmental Processes, 4*, 59-69.
- Müller, M., Maier, K., & Kals, E. (2013). Klimaschützendes Handeln im Haushalt: Die Rolle von emotionaler Bindung an die Natur. *Umweltpsychologie*, *17*(1), 60-73.
- Munder, T., Flückiger, C., Leichsenring, F., Abbass, A. A., Hilsenroth, M. J., Luyten, P., Rabung, S., Steinert, C., & Wampold, B. E. (2019). Is psychotherapy effective? A re-analysis of treatments for depression. *Epidemiology and Psychiatric Sciences*, 28(3), 268–274. https://doi.org/10.1017/S2045796018000355
- Neff, K. D., Tóth-Király, I., Knox, M. C., Kuchar, A., & Davidson, O. (2021). The development and validation of the state self-compassion scale (long-and short form). *Mindfulness*, *12*, 121-140. https://doi.org/10.1007/s12671-020-01505-4
- Nicolè, S., & Seeland, K. (1999). Die sozialintegrativen Wirkungen von Parks und Wäldern als gestaltete Naturräume. Erste Ergebnisse zweier Untersuchungen in der Schweiz und in Deutschland. *Schweizerische Zeitschrift für Forstwesen, 150*, 362-369.
- Olafsdottir, G., Cloke, P., Schulz, A., van Dyck, Z., Eysteinsson, T., Thorleifsdottir, B., & Vögele, C. (2020). Health Benefits of Walking in Nature: A Randomized Controlled Study Under Conditions

of Real-Life Stress. *Environment and Behavior, 52*(3), 248–274. https://doi.org/10.1177/0013916518800798

- Owens, M., & Bunce, H. L. I. (2022). Nature-Based Meditation, Rumination and Mental Wellbeing. International Journal of Environmental Research and Public Health, 19(15). https://doi.org/10.3390/ijerph19159118
- Peters, A., Rospleszcz, S., Greiser, K. H., Dallavalle, M., & Berger, K. (2020). Covid-19-Pandemie verändert die subjektive Gesundheit. Erste Ergebnisse der NAKO-Gesundheitsstudie. *Deutsches Ärzteblatt International*, 117(50), 861-867.
- Petzold, H. G., Ellerbrock, B., & Hömberg, R. (2019). Die "Neuen Naturtherapien": Formen, Konzepte, Perspektiven – eine Übersicht. In H. G. Petzold, B. Ellerbrock, & R. Hömberg (Eds.), Die Neuen Naturtherapien: Handbuch der Garten-, Landschafts-, Wald- und Tiergestützten Therapie (pp. 31–70). Aisthesis Verlag.
- Polz-Watzenig, A. (2020). *Die heilsame Wirkung des Waldes in der Integrativen Therapie.* Springer. https://doi.org/10.1007/978-3-658-30670-0
- Rogers, C. R. (1981). Der neue Mensch. Klett-Cotta.
- Russell, R., Guerry, A. D., Balvanera, P., Gould, R. K., Basurto, X., Chan, K. M. A., Klain, S., Levine, J., & Tam, J. (2013). Humans and Nature: How Knowing and Experiencing Nature Affect Well-Being.
 Annual Review of Environment and Resources, 38(1), 473–502.
 https://doi.org/10.1146/annurev-environ-012312-110838
- Shanahan, D. F., Astell-Burt, T., Barber, E. A., Brymer, E., Cox, D. T. C., Dean, J., Depledge, M., Fuller, R. A., Hartig, T., Irvine, K. N., Jones, A., Kikillus, H., Lovell, R., Mitchell, R., Niemelä, J., Nieuwenhuijsen, M., Pretty, J., Townsend, M., van Heezik, Y., ... Gaston, K. J. (2019). Nature-Based Interventions for Improving Health and Wellbeing: The Purpose, the People and the Outcomes. *Sports*, 7(6), 1-20. https://doi.org/10.3390/sports7060141
- Steigen, A. M., Kogstad, R., & Hummelvoll, J. K. (2016). Green Care services in the Nordic countries: An integrative literature review. *European Journal of Social Work*, 19(5), 692–715. https://doi.org/10.1080/13691457.2015.1082983
- Steinke, I. (1999). Kriterien qualitativer Forschung. Ansätze zur Bewertung qualitativ-empirischer Sozialforschung. Juventa.
- Tennessen, C. M., & Cimprich, B. (1995). Views to nature: effects on attention. *Journal of Environmental Psychology*, *15*, 77-85.
- Thom, J., Bretschneider, J., Kraus, N., Handerer, J., & Jacobi, F. (2019). Versorgungsepidemiologie psychischer Störungen. Bundesgesundheitsblatt- Gesundheitsforschung-Gesundheitsschutz, 62(2), 128-139. https://doi.org/10.1007/s00103-018-2867-z

- UNESCO-Biosphärenreservat Rhön. (2018). *Neues Rahmenkonzept 2018. UNESCO-Biosphärenreservat Rhön. Band III Wie sieht unser Weg aus?*. https://www.biosphaerenreservat-rhoen.de/fileadmin/media/publikationen/Rahmenkonzept_Band_III.pdf
- Ulrich, R. S. (1983). Aesthetic and affective response to natural environment. In I. Altman, & J. F. Wohlwill (Eds.), *Behavior and the Natural Environment* (pp. 85-125). Plenum Press.
- Ulrich, R. S. (1984). View through a window may influence recovery from surgery. *Science*, 224(4647), 420-421. https://doi.org/10.1126/science.6143402
- Ulrich, R. S., Simons, R. F., Losito, B. D., Fiorito, E., Miles, M. A., & Zelson, M. (1991). Stress recovery during exposure to natural and urban environments. *Journal of Environmental Psychology*, 11(3), 201-230. https://doi.org/10.1016/S0272- 4944(05)80184-7
- Wastlhuber, L. (2019). Naturschützende Engagementbereitschaften und Verhaltensweisen: Eine psychologische Studie zu Naturerfahrungen, emotionaler Verbundenheit und Achtsamkeit in der Natur [master's thesis]. Katholische Universität Eichstätt-Ingolstadt.
- Watson, D., Clark, L. A., & Tellegen, A. (1988). *Positive and negative affect schedule*. American Psychological Association.
- Weinstein, N., Przybylski, A. K., & Ryan, R. M. (2009). Can Nature Make Us More Caring? Effects of Immersion in Nature on Intrinsic Aspirations and Generosity. *Personality and Social Psychology Bulletin*, 35(10), 1315–1329. https://doi.org/10.1177/0146167209341649
- Wilson, E. O. (1984). Biophilia. Harvard University Press.
- World Health Organization (WHO) (1948, April 7). *Constitution of the World Health Organization*. WHO. https://www.who.int/governance/eb/who_constitution_en.pdf
- World Health Organization (WHO) (1986, November 21). Ottawa-Charter for Health Promotion. First International Conference on Health Promotion. WHO. https://www.euro.who.int/__data/assets/pdf_file/0004/129532/Ottawa_Charter.pdf
- Zieris, P., Freund, S, & Kals, E. (2023). *Nature experience and well-being: Evaluation of bird watching as a preventive intervention in nursing homes* [Manuscript submitted for publication].